ABSTRACT

Title of Document:

CHARACTERIAZTION OF ELECTRICALLY ACTIVE DEFECTS IN ADVANCED GATE DIELECTRICS

Dawei Heh, Ph. D., 2005

Directed By:

Professor Joseph B. Bernstein, Department of Electrical and Computer Engineering

As the gate oxide thickness of the metal-oxide-semiconductor (MOS) Field Effect Transistor (FET) is continuously scaled down with lateral device dimensions, the gate leakage current during operation increases exponentially. This increase in leakage current raises concerns regarding device reliability. Substitute dielectrics with high dielectric constant (high-k) have been proposed to replace traditional SiO₂ to reduce the leakage current in future devices. However, these high-k dielectrics also have reliability issues due to the large amount of intrinsic trapping centers.

In this work, electrically active defects generated during electrical stress of ultrathin SiO_2 dielectrics are characterized and studied. The mechanism of oxide breakdown is studied by investigating the contributions of hot holes to device time-tobreakdown (t_{bd}). The proper extrapolation of t_{bd} from accelerated testing conditions to normal device operating conditions is also studied. The factors that affect this extrapolation are discussed. Another important device reliability parameter, threshold voltage shift (ΔV_{th}), is also investigated in this work. The dominant mechanisms causing this shift is studied using both simulation and experimental results.

The current primary reliability issue with high-k dielectrics is the large amount of intrinsic traps located in the dielectric stack. Therefore, the electrical characterization of high-k dielectrics in this work is focused on these initial asfabricated trapping centers. A methodology based on 2-level charge pumping (CP) measurements at different frequencies is used to study the spatial profile of these trapping centers. The correlation between device fabrication data and measurement results indicates this methodology is accurate and reliable.

CHARACTERIZATION OF ELECTRICALLY ACTIVE DEFECTS IN ADVANCED GATE DIELECTRICS

By

Da-Wei Heh

Dissertation submitted to the Faculty of the Graduate School of the University of Maryland, College Park, in partial fulfillment of the requirements for the degree of Doctor of Philosophy 2005

Advisory Committee: Professor Joseph B. Bernstein, Chair/Advisor Professor Martin C. Peckerar Professor Chia-Hung Yang Professor Carol Smidt Dr. Eric M. Vogel, Co-Advisor © Copyright by Da-Wei Heh 2005

Acknowledgements

This work cannot be done without the generous support from my academic advisor, Professor Bernstein and my research advisory, Dr. Vogel, at NIST. I would like to thank them for their guidance and advice in the past four years. I would also like to thank Professor Peckerar, Professor Yang, and Professor Smidt for serving on my advisory committee.

I want to thank Dr. Suehle at NIST for his laboratory assistance and helpful discussions throughout my project. It is also a privilege and a pleasure to work with all of the group members the CMOS and Novel Devices Group at NIST. I want to thank them for their help and support.

Table of Contents

Acknowled	gements.		ii
Table of Co	ntents		iii
List of Figu	res		vi
Chapter 1	l. Intr	oduction	1
1.1	Introduc	stion	1
1.2	Purpose	and Approach	5
1.3	Preview	of Thesis	7
Chapter 2	2. Bac	kground	13
2.1	Overvie	w	13
2.2	Percolat	ion Theory	13
2.3	Defect (Generation Models	14
	2.3.1	Thermal-Chemical Electric Field Model	
	2.3.2	Anode Hole Injection Model	15
	2.3.3	Hydrogen Release Model	16
2.4	Statistic	s and Characterization Techniques	17
2.5	Weibull	Statistics	18
2.6	Charact	erization techniques	20
	2.6.1	SILC Measurement	20
	2.6.2	Charge Pumping Measurements	21

Chapter	3. Substrate Hot Hole Injection Experiment	36
3.1	Overview	37
3.2	Introduction	37
3.3	Experiment	37
3.4	Conclusion	40
Chapter	4. Sigmoidal Defect Generation	46
4.1	Overview	46
4.2	Introduction	47
4.3	Determination of Defect Generation Rates	48
4.4	Discussion of Possible Mechanisms for the Nonlinear Behavior of De	fect
	Generation	50
4.5	Conclusion	53
Chapter	5. The Effect of Mobility Degradation to Threshold Volt	age
	Shifts for Ultrathin Gate Oxides	66
5.1	Overview	66
5.2	Introduction	67
5.3	V_{th} and V_{FB} Shift Measurements	68
5.4	Simulation of V _{th} Shift	69
5.5	Conclusion	71
Chapter	6. Electrical Characterization of Spatial Distributions	of
	Trapping Centers in HfO ₂ /SiO ₂ Stacked Dielectrics	77
6.1	Overview	. 77

iv

6.2	Introduction	78
6.3	Simulation of the Probable Range in CP Measurement	79
6.4	Experiment	
6.5	Results and Discussion	
6.6	Conclusion	89
Chapter 7	7. Summary and future Work	97
7.1	Overview	97
7.2	Summary	97
7.3	Future Work	99
Reference	<u>)</u>	100

v

List of Figures

Figure 1.1	The extension of Moore's law made by Intel. It is predicted that the number of transistors per integrated circuit will reach one billion by the year 2005 [4]
Figure 1.2	A plot from the 2004 ITRS showing future projected EOT and limit of gate leakage current density [10]16
Figure 1.3	The gate tunneling current for SiO_2 with $t_{ox} \leq 4nm$. The tunneling current increases exponentially with reducing oxide thickness [13]
Figure 1.4	The comparison of gate leakage current density between SiO_2 and HfO_2 gates at different EOT. The leakage current is significantly reduced for HfO2 gate dielectrics when EOT is reduced [26]
Figure 2.1	Schematic illustration of oxide breakdown proposed by percolation theory [34]. As indicated, the open circles represent the randomly generated electrical defects during stress. When these defects electrically connect the poly-Si and Si-SiO ₂ interfaces (shaded circles), they cause oxide breakdown
Figure 2.2	Illustration of AHI model for n-channel MOSFET. Energetic electrons are injected from cathode (n-channel) and then tunnel into anode (n+ poly-Si). Holes are generated at anode and tunnel back into oxide due to gate bias
Figure 2.3	The time-to-breakdown comparison between devices with H ₂ annealing and D ₂ annealing [68]
Figure 2.4	Illustration of the general statistical functions. (a) cumulative distribution function (CDF), (b) reliability function [72]30
Figure 2.5	Example of statistic Weibull distribution plot of t_{BD} . The slope of the curve is the shape parameter, β . The t_{BD} value of which logarithm value is corresponding to Ln(-Ln(1-F)) = 0 (or F=63%) is usually defined as the time-to-breakdown of the system
Figure 2.6	Comparison of Weibull distribution and lognormal distribution. Both distributions have been used to fit experiment data but Weibull can better fit experiment data, especially at low percentile [79]32

Figure 2.7	The experimental setup for CP measurement. The ac bias is applied to the gate while the source and drain can be grounded or applied small reverse bias. The CP current is measured from the grounded substrate [86]
Figure 2.8	The measured I_{CP} as a function of the ramped V_{base} and the relative position of the pumping signal respect to the device V_{FB} and V_{th} . The averaged interface state density is proportional to the I_{CP} measured in region 1 [86]
Figure 2.9	The energy distribution of interface trap density for n-channel MOSFET measured by CP method with varying t_r and t_f [84]35
Figure 3.1	Schematic illustration of device structure in this work42
Figure 3.2	Characteristic interface defect generation (CP data) under SHH injection and subsequent CVS
Figure 3.3	Weibull distribution function versus injected-hot-hole-charge-to- breakdown (Q_{bd}^{hh}) under SHH stress
Figure 3.4	Weibull distributions of Q_{bd} for devices under CVS with different amount of pre-injected hot holes by using SHH stress. The bars indicate the 95% confidence intervals for modal Q_{bd} s45
Figure 4.1	Extrapolation of t_{BD} to device normal operating condition (low stress voltage) from accelerated stress condition (high stress voltage). It shows different extrapolations can cause the variations in predicting device lifetime (t_{BD}) as large as several orders of magnitude [114]
Figure 4.2	Illustration of experimental setup for the pulse stress experiment55
Figure 4.3	Defect generations at different frequencies under pulse stress. The overlaps of these curves showed that this technique can be used to measure defect density at a large range of fluence
Figure 4.4	Illustration of linear regression fitting curve of defect generation at certain stress voltage
Figure 4.5	Comparison of linear regression fitting curves of defect generations at different stress voltages
Figure 4.6	The comparison of linear regression fitting curves of P _g at different stress voltages

Figure 4.7	Comparison of voltage accelerations of P_g in linear regime, saturated regime and the reciprocal of Q_{inj}
Figure 4.8	Comparison of defect generation of de-trapping model, hydrogen diffusion model and experimental data
Figure 4.9	Extrapolated carrier capture cross section change over injected fluence
Figure 4.10	Extracted averaged interface state density change over injected fluence
Figure 4.11	Comparison of CP data curve (σ is assumed constant) and D _{it} generation curve (σ is changing with Q _{inj}) with linear defect generation curve
Figure 4.12	Comparison of carrier capture cross sections over injected fluence: experimentally extrapolated (σ) and calculated from linear defect generation assumption (σ ')
Figure 5.1	Illustration of extraction method of V_{FB} shifts for devices measured by using the C-V method. The insert is the complete C-V curve, and the figure showed the magnified part around V_{FB}
Figure 5.2	Comparison of V_{th} shifts measured by I_d - V_g and V_{FB} shifts measured by C-V methods for n-type MOSFET stressed in inversion73
Figure 5.3	Comparison of V_{th} shifts measured by I_d - V_g and V_{FB} shifts measured by C-V methods for n-type MOSFET stressed in inversion74
Figure 5.4	Simulation results of V _{th} shifts in n-type MOSFET at different trap generation conditions
Figure 5.5	Simulation results of V _{th} shifts in p-type MOSFET at different trap generation conditions
Figure 6.1	Illustration of the change of F value with respect to a square pulse
Figure 6.2	3-D ΔF contour simulation result. ΔF is equal to one within the trapezoidal plateau that indicates the region having the maximum probability been probed
Figure 6.3	Simulation results of pulse parameters dependence of ΔF contour. In (a) both t _{on} and t _{off} are equal to 50 ns, while in (b) both t _{on} and t _{off} are equal

Chapter 1

Introduction

1.1 Introduction

Ever since the first transistor was introduced by Shockley and Pearson in 1948 [1], the device with metal, oxide and semiconductor stacked structure has dramatically affected and changed the world. Today, the Metal-Oxide-Semiconductor-Field-Effect-Transistor (MOSFET) is the basic building block of modern integrated circuits and can be found everywhere from kitchen appliances to the space shuttle. The remarkable capability of the MOSFET comes from the voltagemodulated conductance of the semiconductor surface underneath the oxide layer. When a gate voltage is applied to the metal layer (or highly doped poly-silicon layer), minority carriers from the source and the drain are either accumulated or depleted at the semiconductor surface. This drives the device into the "on" and "off" states, accordingly. The key issue of this voltage-controlled characteristic of the MOSFET is attributed to the existence of the oxide layer sandwiched between the metal and the semiconductor layers. The oxide layer serves as an insulation layer by blocking the current between the gate electrode and the semiconductor when mobile carriers are accumulated at the semiconductor layer surface.

Silicon dioxide (SiO_2) has been used as this insulation material for more than three decades [2]. The success of the Si-SiO₂ system is because of the unique features of SiO₂ such as high band gap energy, high thermal stability, and the excellent compatibility with the Complementary MOS (CMOS) technology. Furthermore, the high quality interface between the thermally grown SiO₂ and Si substrate makes the Si-SiO₂ system possess relatively high electron and hole mobilities and low interface states. These outstanding properties make the Si-SiO₂ system irreplaceable in the past decades.

Because of the improvement of process integration technology, the device density of integrated circuits has doubled for approximately every two years, according to Moore's Law [3]. As shown in Figure 1.1, Intel has predicted that the number of transistors per integrated circuit will reach one billion by the year 2005 [4]. As the device density of the integrated circuits increases, all of the device dimensions are scaled downward. As device dimensions are scaled, the operating voltage, V_{DD}, is also scaled by the same factor as the device dimensions. The substrate doping density must then be increased in order to reduce short-channel effects. To keep the driving current for proper circuit operation, the gate oxide capacitance (C_{ox}) must be increased [5]. Historically, this has been accomplished through a reduction of the gate oxide thickness (t_{ox}) [5, 6]. The 2004 International Technology Roadmap for Semiconductors (ITRS) predicts that the equivalent gate dielectric thickness will be reduced to approximately 7 Å by 2010, as shown in Figure 1.2 [7]. With the scaling of t_{ox} , SiO₂ will no longer be able to block leakage current effectively. The exponentially increasing gate leakage current with decreasing t_{ox} has been considered as the essential factor limiting the future scaling of t_{ox} . The gate leakage current will increase the device standby power consumption [8] and also affect the circuit performance [9]. For $t_{ox} \le 4$ nm, it is found that the gate leakage current increases by approximately one order of magnitude for every 0.2-0.3 nm of oxide thickness reduction [10], as shown in Figure 1.3. The maximum tolerable gate leakage current is suggested to be between 1 (A/cm²) and 10 (A/cm²), which correspond to an oxide thickness between 1.2 and 1.5 nm [11-13].

The gate leakage current not only causes a power dissipation problem, but also raises concerns regarding the reliability of the gate oxide. It is known that electrically active defects are generated when electrons or holes tunnel through the gate oxide [6,9-12]. These defects cause shifts in V_{th} and degrade the channel carrier mobility impacting the device performance. When the generated defects reach a critical amount, they cause the gate oxide to breakdown. Although the reliability of ultra-thin SiO₂ has been heavily studied, the exact physical mechanisms of defect generation of breakdown are still unknown.

To reduce the large gate leakage with the scaling oxide thickness, substitute materials with high dielectric constant (high-k), such as Al_2O_3 , HfO_2 and ZrO_2 , have been proposed to replace SiO₂ [14-22]. The principle of replacing SiO₂ with these high-k materials is to keep C_{ox} the same while permitting a physically thicker dielectric. C_{ox} for SiO₂ and high-k materials can be expressed as

where ε_0 is the vacuum dielectric constant, ε_{SiO_2} and ε_k are the relative dielectric constants for SiO₂ and high-k dielectrics respectively, while t_{SiO_2} and t_k represent the physical thickness of SiO₂ and high-k dielectrics respectively. In order to compare high-k dielectrics with SiO₂, equivalent oxide thickness (EOT) is defined for high-k dielectrics as the equivalent thickness of SiO₂ which results in the same capacitance as the high-k. From equation (1.1) it is easy to show that EOT can be expressed as

Since the typical value for ε_k is between 10 and 25, which is larger than ε_{SiO_2} (=3.9), the physical thickness of high-k dielectrics can be relatively thicker (~ 3 nm) while keeping EOT as small as 1 nm. Because the physical thickness of high-k materials is much thicker than SiO₂ for a given capacitance, the tunneling current is expected to be reduced significantly for high-k gate dielectrics [23, 24]. Figure 1.4 shows the comparison of the gate leakage current density (J_g) between SiO₂ and HfO₂ gates at different EOT. It is obvious that J_g is reduced significantly for the HfO₂ gate dielectric as compared to SiO₂.

The search of proper high-k dielectrics to replace SiO_2 has proven to be extremely challenging. Problems associated with these dielectrics include thermal instability on silicon, instability with gate electrode materials, large tunneling currents, and lower than expected dielectric constants of the deposited films. Furthermore, unlike the $Si-SiO_2$ system which shows a high quality interface with low interface state density and high channel carrier mobilities, the Si-high-k system generally has large interface state density and low channel carrier mobilities. [25-30].

To obtain a high quality interface with Si substrate while using a substitute high-k gate dielectric, a SiO₂-like interfacial layer can be grown before the high-k dielectric is deposited. This interfacial layer provides a transition region from the Si substrate to the substitute high-k dielectrics and is required to form high quality interfaces with both Si substrate and high-k dielectrics. This high-k and interfacial layer stacked gate structure makes the high-k dielectric system even more complicated to characterize.

There has been a lot of recent research related to high-k dielectrics [14-22]. The main issue related to the high-k gate dielectrics themselves is the large amount of fixed charges and charge trapping centers compared with SiO₂ [31]. The charge trapping centers are believed to exist at the interface with gate electrode [32], the interface with SiO₂-like interfacial layer [33] and inside the bulk dielectric. These trapping centers and intrinsic charge defects make high-k dielectrics much more complicated than the traditional SiO₂ gate dielectric.

1.2 Purpose and Approach

The overall goal of the research presented in this thesis is to provide detailed electrical characterization and fundamental understanding of electrically active defects in advanced gate dielectrics, including the traditional ultrathin SiO_2 and HfO_2 dielectrics. The detailed approach to achieve this goal is described in the following.

The initial work involved the study of MOSFETs with ultrathin SiO₂. Since SiO_2 is relatively free of initial defects, the focus was on defect generation mechanisms. Proposed mechanisms of defect generation and oxide breakdown from the literature were studied by injecting different carriers (electrons and holes) during stress. The behavior of the time-to-breakdown (t_{bd}) and charge-to-breakdown (Q_{bd}) of the devices was used to compare and eliminate less likely mechanisms. Attention was then turned to the relation between the defect generation and the injected charges by using an interrupted-stress method. The understanding of how to perform proper extrapolation and the correct interpretation of the data from electrical characterization results was the focus of this study.

The understanding and characterization methodologies developed in ultrathin SiO_2 were then applied to MOSFETs with HfO₂ gate dielectrics. Since the main problem associated with HfO₂ is the relatively large amount of initial defects, the focus of this research was to understand how to properly characterize these initial defects. The initial defect characterization was conducted through measurements of the spatial as well as the energy distributions of defects in HfO₂ using charge-pumping. By changing the measurement conditions, defects at different energies and depths in HfO₂ can be characterized.

1.3 Preview of Thesis

This thesis is organized into seven chapters.

Chapter 2 provides background information for the rest of the thesis. It starts with the introduction of the percolation theory that connects the microscopic defect generation to the macroscopic gate oxide degradation phenomena and the eventual breakdown. The three major models proposed in the literature to explain defect generation mechanisms are then introduced. The success and controversies of these models are discussed in detail. The statistics of breakdown is then introduced followed by the description of the electrical characterization techniques such as stress induced leakage current (SILC) and charge-pumping (CP) measurements that are used to monitor the dynamic defect generation during a stress.

Chapter 3 describes my research on possible defect generation models for SiO_2 by studying the effect of injected substrate hot holes on defect generation and t_{bd} . It is found that pre-injected substrate hot holes have no effect on the t_{bd} for the subsequent constant voltage stress (CVS). The results suggest that holes are not responsible for defect generation and breakdown during CVS. The results also suggest that different types of defects may be generated during the CVS as compared to substrate hot hole injection.

Chapter 4 describes defect generation mechanisms in SiO_2 by inspecting defect generation as a function of injected charges. A non-linear relationship between the generated defects and the injected charges are revealed. This non-linear relationship raises the concern of extrapolating the t_{bd} from stress conditions (high voltage) to device normal operating conditions (low voltage) correctly. The possible

mechanisms that cause this non-linear relationship are also investigated and it is found that the changes of carrier capture cross sections during stresses can not explain this non-linear relationship.

Chapter 5 investigates the dominant mechanism that causes the threshold voltage (V_{th}) shifts in n- and p-channel MOSFETs. The results suggest that mobility degradation can be an important component of threshold voltage shift. The results suggest that proposed oxide degradation models based on the I_d - V_g measurements of V_{th} may not be accurate.

In Chapter 6, the focus is changed to the HfO_2 gated MOSFET. The spatial and energy distributions of the initial defects in HfO_2 are characterized by using CP. The fraction of defects probed by CP at a given energy and depth within the dielectric is simulated. By changing CP measurement conditions, the spatial and energy profiles of defects in HfO_2 can be characterized. The simulation results are compared with experimental data to extract the initial defect distribution in the HfO_2 dielectric.

Finally, Chapter 7 provides the summary of this thesis and some possible future work.



Figure 1.1 The extension of Moore's law made by Intel. It is predicted that the number of transistors per integrated circuit will reach one billion by the year 2005 [4].



Figure 1.2 A plot from the 2004 ITRS showing future projected EOT and limit of gate leakage current density [10].



Figure 1.3 The gate tunneling current for SiO_2 with $t_{ox} \leq 4nm$. The tunneling current increases exponentially with reducing oxide thickness [13].



Figure 1.4 The comparison of gate leakage current density between SiO_2 and HfO_2 gates at different EOT. The leakage current is significantly reduced for HfO2 gate dielectrics when EOT is reduced [26].

Chapter 2

Background

2.1 Overview

The goal of this chapter is to provide the background that will be helpful to read the rest of this thesis. Macroscopic gate oxide degradation phenomena and eventual breakdown are first explained through percolation theory. The three major proposed empirical models for defect generation are discussed in detail, including the success and controversies of these models. The statistics of breakdown and electrical characterization techniques such as stress induced leakage current (SILC) and charge pumping (CP) will then be introduced.

2.2 Percolation Theory

Gate oxide degradation and eventual breakdown are due to the generation of defects during stresses. To explain how electrical defects cause oxide breakdown, Degraeve proposed the percolation theory in 1995 [34]. It suggests that electrical defects are generated randomly in the gate oxide during a stress, as the open circles

schematically depict in Figure 2.1. The energy levels of these defects are located within the silicon band gap and thus provide additional intermediate tunneling paths for electrons and holes. These tunneling paths degrade the oxide insulation ability and increase gate leakage current. As the defects are continuously generated, they have a chance to connect to each other electrically and form a conducting path. Once the conducting path connects the poly-Si gate to the Silicon substrate (as the shaded circles show in Figure 2.1), an electrical short across the oxide is found. This conducting path surges a large current that causes permanent structural breakdown of the oxide.

Percolation theory successfully explains how the defects degrade and cause eventual oxide breakdown. The number of defects at breakdown (N_{BD}) and the effective defect size (or the effective number of defects that causes the breakdown) can also be calculated by fitting this theory with experimental data [34-36]. The theory also successfully explains the dependence of N_{BD} on oxide thickness and area. However, percolation theory does not explain how these defects are generated.

2.3 Defect Generation Models

To explain how the defects described in percolation theory are generated, several physical models have been proposed in the literature. Three major proposed models, the thermo-chemical electric field model, the anode hole injection (AHI) model and the hydrogen release (HR) model, will be introduced in the following sections.

2.3.1 Thermo-Chemical Electrical Field Model

The thermo-chemical electric field model was developed based on the observation of t_{bd} as a function of electrical field [37-39]. It suggests that defects are generated due to the interaction between the electric field and the oxide lattice [40]. When an electric field is applied across the SiO₂ layer, it interacts with the weak Si-Si bonds associated with oxygen vacancies in the SiO₂ layer. The applied electric field eventually breaks the weak bonds and creates permanent defects which lead to breakdown [41, 42]. This electric field driven model is no longer accepted as recent studies conclusively showed that defect generation is related to electron fluence and cannot be explained by the interaction of the SiO₂ lattice with electrical field [43-46].

2.3.2 Anode Hole Injection Model

The Anode Hole Injection (AHI) model suggests that hot holes are generated at the anode due to energetic electrons injected from the cathode. These holes can then tunnel back into the oxide and generate defects that cause gate oxide degradation [47-49]. Figure 2.2 shows illustratively the processes of the energetic tunneling electron, the anode hole generation, and the anode hole injection for an n-channel MOSFET. The oxide breakdown happens when a critical hole fluence (Q_p) is reached [50]. The AHI model was later modified to include anode holes generated through minority ionization, which makes the anode hole generation possible at the low gate bias condition [51, 52].

The AHI model is based on experimental observations of thicker oxides that the number of defects generated in the gate oxide is uniquely correlated to the anode hole fluence, independent of the gate stress voltage and the oxide thickness [50]. The hole fluence is assumed to be equal to the measured substrate current which is proportional to the gate current during stress [53]. The AHI model can also explain the polarity dependence observed from both n-channel and p-channel devices stressed at opposite polarity bias conditions [54-56]. This model is supported by the observation that Q_p is independent of the gate stress voltage (V_G) and t_{ox} [53]. The AHI model also shows that the voltage acceleration factor (γ), which is defined as the negative derivative of the logarithm of t_{bd} respect to the stress voltage (- $\partial \ln t_{bd}/\partial V$), should increase with decreasing gate voltage, as observed from experiment [57].

Nevertheless, the AHI model is challenged by other observations. First, Q_p is found not to be a constant for low electron injection conditions [58]. It is also found that Q_p decreases with the decreasing oxide thickness [49], and shows a temperature dependence [59, 60]. Furthermore, physical mechanisms other than anode hole injection can also contribute to the measured substrate current [61, 62]. Other observations also indicate that even though the injected anode holes are efficient in generating electrical defects, these defects are inefficient in causing oxide breakdown [60].

2.3.3 Hydrogen Release Model

The other model that tries to explain defect generation is the hydrogen release (HR) model. The HR model is similar to the AHI model but suggests that hydrogen species are generated at the anode by energetic electrons and do damage when they drift or diffuse into the gate oxide [60]. It is suggested that atomic

hydrogen can attack Si-O bonds and cause the damage. Hot electrons can also break Si-H bonds, which are believed to be one of the sources of oxide degradation [57, 63, 64]. It is also found that the voltage dependence of the H atom desorption rate from the silicon surface is similar to the voltage dependence of the defect generation rate (P_g) in the gate oxide [65]. The HR model can also possibly explain the large exponent value ($n \sim 44$) in the t_{bd} power law dependence of the stress voltage [65-67].

The HR model has been questioned because some results suggest that Q_{bd} is not improved if the deuterium is used to passivate the Si-SiO₂ interface [68], as shown in Figure 2.3. It has been reported that deuterated oxide films have suppressed hydrogen desorption from the Si interface and, therefore, improved the immunity to the interfacial trap generation during channel hot carrier injection [69]. Nevertheless, it is also reported from other groups that deuterated oxide does improve device reliability [70, 71]. Therefore, the debate remains.

2.4 Statistics and Characterization Techniques

Percolation theory explains the oxide degradation process in terms of the defect generation from a microscopic point of view. To study how defects are generated, however, we can only rely on either observed statistical phenomenon or macroscopic electrical characteristics to correctly analyze the experimental data and link them to the defect generation mechanism. To analyze statistical phenomenon such as time-to-breakdown (t_{bd}) (also known as time-dependent-dielectric-breakdown TDDB) and charge-to-breakdown ($Q_{bd} \approx t_{bd} \cdot J_g$, where J_g is the current density during a stress), proper statistics is necessary; while for characterizing electrical properties

such as interface state density (N_{it}) generation and gate tunneling current increase, electrical characterization techniques are needed. In the following sections, available statistics and electrical characterization techniques that are used to study the defect generation mechanisms from different aspects will be introduced separately.

2.5 Weibull Statistics

Percolation theory proposes that the formation of an electrical conducting path triggers oxide breakdown. However, since the formation of a percolation path is a random process, N_{BD} (or macroscopically, Q_{bd} and t_{bd}) for different devices can vary by orders of magnitude. Therefore, statistics is required to describe the failure distribution and define N_{BD} as well as t_{bd} quantitatively.

Consider a system with N identical devices under the same electrical stress condition. The total number of failed devices at any given time can be defined as $N_f(t)$ and the ratio of $N_f(t)$ to N is defined as the cumulative distribution function (CDF), F(t). Figure 2.4 (a) shows the general behavior of F(t). It is noticed that F(t) is zero initially and goes to one when time approaches to infinity. The counterpart function of F(t) is the reliability function R(t), which is defined as $R(t) \equiv 1$ -F(t), as shown in Figure 2.4 (b). Meanwhile, R(t) can also be mathematically expressed as

where β is called the shape parameter and θ is called the scale parameter [72].

By using the definition of R(t) and equation 2.1, the following expression can be derived,

where ln(-ln(1-F(t))) is the Weibull distribution function. By plotting the Weibull distribution function with respect to the natural logarithm of the t_{bd} of the N devices, the slope of the curve is equal to β . The scale parameter θ , which corresponds to 63% of the failure, is the modal value of the t_{bd} of the N devices and can be defined as the t_{bd} of this N-device system, as shown in Figure 2.5. Similarly, the functions F(t) and R(t) could be defined as functions of stress injected charge instead of stress time and the Weibull distribution for Q_{bd} can then be obtained.

The β value of the Weibull distribution indicates the spread of the t_{bd} distribution. It is found that oxide breakdown caused by extrinsic defects (non-stress related) and intrinsic defects (stress related) would result in different β values [73]. Therefore, it is possible that β value can be used to identify different defect species generated by stress. On the other hand, the modal value of t_{bd} from the Weibull distribution is directly affected by the stress conditions. Therefore, by comparing the β value and the modal value of t_{bd}, the effect of gate oxide fabrication process and applied stress conditions on breakdown can be revealed.

Historically, the lognormal distribution [78] has also been used to fit experimental data to describe t_{bd} distribution. However, recent reports have indicated that that Weibull distribution can better fit t_{bd} data for large number of samples. This permits more accurate extrapolation of t_{bd} to low percentile values [74, 75, 79-81], as shown in Figure 2.6.

2.6 Characterization Techniques

Another strategy for studying defect generation mechanisms is characterizing electrically active defects during a stress. To achieve this goal, it is required to interrupt a stress periodically and perform electrical measurements. These results can then be linked to microscopic defect generation. This interrupted-stress strategy can reveal the relation between stress-induced defects and stress time (or injected electrons and/or holes). The most basic requirement of this strategy is that no further defects are generated during the measurement. Therefore, available measurement techniques must be carefully chosen and measurement conditions should be optimized.

Two techniques commonly used to electrically characterize gate oxide reliability are charge pumping (CP) measurements [83-90] and stress induced leakage current (SILC) [91-95] measurements. The relative change of maximum CP current (I_{CP}) is used to monitor the interface state density at Si-SiO₂ interface [83-85] and the SILC is used to measure the bulk oxide defect density [91-93].

2.6.1 SILC Measurement

During SILC measurements, a sense voltage (V_{sense}) is applied to the gate while all the other terminals of an MOSFET or a capacitor are grounded. V_{sense} should be smaller than the stress voltage to avoid any further stress during the measurement [91]. The gate leakage current measured during SILC measurement is thought to be the electrical conducting path due to the generation of neutral-defect sites [96] or oxygen vacancies [92] inside the bulk oxide. It has been suggested that SILC can be described using the following equation [95],

where J_G is the SILC, N_{TS} is the defect concentration, c_L and c_R are the defect capture rates at the cathode and anode respectively, and f_L and f_R are the Fermi distributions at the cathode and anode respectively. Therefore, by measuring the change of the SILC between successive stresses, the defect generation inside the bulk oxide due to a stress can be explored.

2.6.2 Charge Pumping Measurement

The experimental setup for CP is shown in Figure 2.7 [86]. A periodic ac pumping signal is applied to the gate, and the source and drain are either applied a small reverse bias or grounded. The substrate is also grounded and a dc CP current is measured from the substrate [85]. When performing the CP measurement a trapezoidal wave with the fixed amplitude (ΔV_A) and frequency, which is widely chosen as the pumping signal, is applied to the gate. During the measurement, the lower output voltage level of the trapezoidal wave (V_{base}) as well as the upper output voltage level of the trapezoidal wave (V_{top}) is ramped up, as shown in Figure 2.8. Figure 2.8 shows the typical measured CP current (I_{CP}) for a n-channel MOSFET as a function of V_{base} . The relative position of the gate pumping signal voltage level with respect to the device flat-band voltage (V_{FB}) and threshold voltage (V_{th}) is also shown in the figure and five different characteristic regions can be distinguished during the voltage ramp-up.

In region 1, where V_{top} ($V_{top} = V_{base} + \Delta V_A$) is higher than V_{th} while V_{base} is lower than V_{FB} , a significant dc current from the substrate due to the electron and hole recombination is measured and is denoted as $I_{CP,MAX}$ [83, 84]. This recombination happens because the device was in the accumulation state initially when the voltage applied to the gate was equal to V_{base} , which is lower than V_{FB} . All the interface states are occupied by holes at this moment. When the pulse passes through, the voltage applied to the gate switches to V_{top} . Since V_{top} is higher than V_{th} , the device is turned into the inversion state instantly. Electrons from the source and drain flush into the channel region and fill the interface states while holes are expelled from the channel area. For the holes occupying the interface states, however, they are "trapped" and cannot move back to the substrate during the short transition of the states. As a result, they are recombined with the electrons.

Similarly, after the pulse passed by the voltage applied to the gate came back to V_{base} again and the device went back to the accumulation state with holes refilling the channel region. However, since the electrons trapped the at interface states cannot move back to source or drain so quickly, they are recombined with the holes. These two recombination currents from interface states contribute to the measured dc $I_{CP,MAX}$ [83-85].

In region 2, since both of the V_{base} and V_{top} are lower than the V_{FB} , the interface states are always filled with holes. In region 3, when V_{base} and V_{top} are both

higher than the V_{th} , the interface states are always filled with electrons. As a result, no recombination happens in these two regions and the charge pumping current is zero.

The equation for I_{CP,MAX} in terms of interface state density is [90]:

$$I_{CP,MAX} = q N_{it} f A_G \cdots (2.4)$$

where,

I _{CP,MAX}	is the maximum charge pumping (Amp),
q	is the electron charge (Coul),
N _{it}	is the total interface state density (cm ⁻²),
f	is the pumping signal frequency (s ⁻¹), and
A_{G}	is the device channel area (cm^{-2}) .

Furthermore, Nit can be expressed as

$$N_{it} = \int_{E_1}^{E_2} D_{it}(E) \ dE \cdots (2.5)$$

where $D_{it}(E)$ is the interface state density at energy level E (cm⁻²-eV⁻¹), and E₁ and E₂ are the lower and upper energy limits of the interface state density distribution, respectively [84]. By using equation (2.5), equation (2.4) can be rewritten as

where $\overline{D_{it}}$ is the averaged interface state density and $\Delta \psi_s$ is the energy range of the interface states. It is shown that $\Delta \psi_s$ can be expressed as [83, 84]:

and therefore,

where

E _{em,h}	is the energy level for the trapped holes (eV),
E _{em,e}	is the energy level for the trapped electrons (eV),
Ei	is the silicon intrinsic Fermi level (eV),
k	is the Boltzman constant (eV/K),
Т	is the absolute temperature (K),
$v_{ m th}$	is the thermal velocity (cm ² /s),
σ_p	is the hole capture cross section (cm ²),
$\sigma_{\rm n}$	is the electron capture cross section (cm^2) ,
ni	is the silicon intrinsic carrier density (cm ⁻³),
--------------------	---
t _{em,h}	is the hole emission time (s),
t _{em,e}	is the electron emission time (s),
$E_{F,acc}$	is the Fermi level in the accumulation state (eV),
E _{F,inv}	is the Fermi level in the inversion state (eV),
t _r	is the rise time of the trapezoidal wave (s), and
t _f	is the fall time of the trapezoidal wave (s).

Then finally, I_{CP,MAX} can be expressed as

$$I_{CP,MAX} = 2 \ q \ \overline{D_{it}} \ f \ A_{G} \ k \ T \ \ln(v_{th} \ n_{i} \ \sqrt{\sigma_{p} \sigma_{n}} \ \frac{|V_{FB} - V_{th}|}{|\Delta V_{A}|} \sqrt{t_{r} t_{f}}) \cdot (2.12)$$

From equation (2.12) it can be seen that if $I_{CP,MAX}$ is plotted as a function of $\sqrt{t_r t_f}$, then the slope will be proportional to \overline{D}_{it} , and the intersect with the x-axis will be proportional to the geometrical mean of electron and holes capture cross section ($\sqrt{\sigma_p \sigma_n}$) [87].

Meanwhile, if equation (2.5) is differentiated with respect to $t_{\rm f}$, it is shown that

Since E_1 is independent of fall time [84] and also from equation (2.9), it is found that

Therefore, by using equation (2.13) and (2.14), the interface state density at trap energy level E_2 can be expressed as

Similarly, the defect density at E_1 can be expressed as

Therefore, by measuring $I_{CP,MAX}$ at various t_r and t_f , the energy distribution of interface states can be extrapolated, as shown in Figure 2.9 [84].



Si substrate

Figure 2.1 Schematic illustration of oxide breakdown proposed by percolation theory [34]. As indicated, the open circles represent the randomly generated electrical defects during stress. When these defects electrically connect the poly-Si and Si-SiO₂ interfaces (shaded circles), they cause oxide breakdown.



Figure 2.2 Illustration of AHI model for n-channel MOSFET. Energetic electrons are injected from cathode (n-channel) and then tunnel into anode (n+ poly-Si). Holes are generated at anode and tunnel back into oxide due to gate bias.



Figure 2.3 The time-to-breakdown comparison between devices with H_2 annealing and D_2 annealing [68].



Figure 2.4 Illustration of the general statistical functions. (a) cumulative distribution function (CDF), (b) reliability function [72].



Figure 2.5 Example of statistic Weibull distribution plot of t_{BD} . The slope of the curve is the shape parameter, β . The t_{BD} value of which logarithm value is corresponding to Ln(-Ln(1-F)) = 0 (or F=63%) is usually defined as the time-to-breakdown of the system.



Figure 2.6 Comparison of Weibull distribution and lognormal distribution. Both distributions have been used to fit experiment data but Weibull can better fit experiment data, especially at low percentile [79].



Figure 2.7 The experimental setup for CP measurement. The ac bias is applied to the gate while the source and drain can be grounded or applied small reverse bias. The CP current is measured from the grounded substrate [86].



Figure 2.8 The measured I_{CP} as a function of the ramped V_{base} and the relative position of the pumping signal respect to the device V_{FB} and V_{th} . The averaged interface state density is proportional to the I_{CP} measured in region 1 [86].



Figure 2.9 The energy distribution of interface trap density for n-channel MOSFET measured by CP method with varying t_r and t_f [84].

Chapter 3

Substrate Hot Hole Injection Experiment

3.1 Overview

Recent work has suggested that the thermo-chemical electric field model cannot explain oxide breakdown. However, it is not clear whether anode hole injection or hydrogen release is the likely mechanism. To determine the more likely model that can describe the gate oxide breakdown, the impact of hot holes on gate oxide breakdown is studied by investigating devices under constant voltage stress with different amount of pre-injected substrate hot holes. The results show that oxide breakdown is independent of the amount of those pre-injected hot holes. This suggests that defects generated by hot holes are not directly related to oxide breakdown during constant voltage stress conditions.

3.2 Introduction

The degradation of the silicon dioxide gate dielectric in Metal-Oxide-Silicon Field Effect Transistors (MOSFETs) has been an interesting and important research topic for several decades [97-101]. Understand the gate oxide breakdown mechanism of MOSFETs is becoming more and more important as the oxide thickness scales down. It is known that electrically active defects are generated when the oxide is under either constant current or voltage stress. Oxide breakdown is believed to be triggered when these defects overlap and form a conducting path connecting the two interfaces of gate oxide [102, 103].

To describe the defect generation, three major models have been proposed: the thermo-chemical electric field model, the hydrogen release model, and the anode hole injection model. For the electric field model recent studies have conclusively shown that defect generation is related to electron fluence and can not just be explained by lattice and electric field interaction. As for the hydrogen release (HR) and anode hole injection (AHI) models, the current controversy between them is whether released hydrogen or generated anode hot holes cause oxide breakdown. A strategy to resolve this controversy is verifying the contributions of either one of these two carriers to oxide degradation.

3.3 Experiment

To focus on the effects of hot holes to oxide degradation, the substrate hot hole injection (SHHI) stress method has been previously used [104, 105]. Under this stress condition, holes are injected from a separate p^+ -doped region, which is called

injector, into a p-channel MOSFET substrate. These holes are then accelerated toward the $Si-SiO_2$ interface using a high substrate bias. When they reach the interface, some of these holes can be tunneled or injected into the gate oxide.

It has been shown that the number of defects generated per injected carrier for SHH stress is much higher than constant voltage stress (CVS) by at least one order of magnitude, which suggests that SHHs are much more efficient in generating or activating defects than electrons [104]. However, it was also reported that charge to breakdown (Q_{bd}) for devices under subsequent CVS was not changed by pre-injected SHHs, which suggests that defects generated by holes are not directly linked to oxide breakdown [104]. One possible explanation for this contradiction is the amount of pre-injected SHHs may have been too small to show significant effects. The main purpose of this work is to inject different amounts of SHHs to determine their effectiveness in causing breakdown.

The devices under test in this work are p-channel MOSFETs in a n-well on a p-type substrate as shown in Figure 3.1. The p-type substrate serves as the hole injector. The gate oxide thickness is approximately 3.5 nm and the device area is 2.5 μ m². Stress induced leakage current (SILC) [106, 107] and charge pumping (CP) measurement [108, 109] are used to monitor the number of defects generated inside the bulk oxide and at the Si-SiO₂ interface, respectively. The stress conditions for SHH injection are V_G =-3 V, V_S = V_D = 0, V_{SUB} = 6 V, and V_{inj} = 7 V; while for CVS, the stress conditions are V_G = -5.2 V, V_S=V_D=V_{SUB}=0. All the stresses and measurements are performed at room temperature.

Figure 3.2 shows the characteristic CP data for the device under SHH stress

and subsequent CVS. CVS is applied to the devices right after SHH stress is terminated (within seconds) to reduce the impact of charging/discharging effect of the traps. It can be seen that defects are generated much faster during SHH stress as compared to CVS, which is consistent with previous reports [104]. Previous work has shown that the defects generated during the subsequent CVS are not due to the interaction of low-energy tunneling electrons with trapped holes in ultra-thin silicon [104, 110]. It has also been shown that defect generation is dependent on electron energy but is independent on hot hole energy. This is because electrically active defects generated by holes are created by hole trapping whereas defects generated by electrons are create by the release of hydrogen [104]. The SILC data (not shown) showed results similar to the CP data.

Figure 3.3 shows the Weibull distribution for injected hot-hole-charge-tobreakdown (Q_{bd}^{hh}) taken from a group of devices under SHH stress only. This data is used to determine the density of SHHs to be pre-injected for subsequent CVS. From this data, two SHH injections were chosen to be used in subsequent CVS: a high injection corresponding to 80% device failure under SHH stress and a low injection corresponding to 10% device failure under SHH stress. Devices which did not undergo breakdown during the SHH stress were used in the subsequent CVS. The CVS Q_{bd} for devices with two different amounts of SHHs were compared to the CVS Q_{bd} for fresh devices (without pre-injected SHHs). By choosing these two different SHH injection levels, it is expected to see a large decrease in modal Q_{bd} with increasing injected SHHs due to the large number of defects created by the holes, if hot holes have any direct effects on CVS oxide breakdown. Figure 3.4 shows the Q_{bd} Weibull distributions of devices under CVS with different amount of pre-injected hot holes. The confidence intervals of modal values for each Q_{BD} are also calculated and shown [111]. It can be clearly seen that there is no significant change in either Weibull slopes or modal values in these distributions, which suggests that defects generated by hot holes are very inefficient in causing breakdown during CVS.

The results suggest that although hot holes are efficient in generating defects, these defects are inefficient in causing breakdown during CVS. However, previous reports [104] have shown that breakdown occurs during SHH stress within limited stress time and with a high N_{BD} . To explain this apparent contradiction, one speculation is that different defects are generated under CVS and SHH injection and these defects do not interact with each other. An observation that the Weibull distribution slope for devices under SHH stress is higher than CVS (comparing Figures 3.3 and 3.4), also suggests that different breakdown mechanisms may dominate or different defects are generated under these two different stress conditions. It is clear that if different defects are generated during SHH stress and CVS, it should be expected to see different Weibull slopes and N_{BD} . If these defects do not interact with each other, it should be expected to see the ineffectiveness of pre-injected SHHs to Q_{bd} for subsequent CVS, as reported in this section.

3.4 Conclusion

The results presented here clearly show that defects generated by hot holes have little or no influence on breakdown during CVS. This has important implications in determining the correct physical model for oxide breakdown. Future work will provide further insight as to why defects generated/activated by SHH stress do not cause oxide breakdown during CVS.



Figure 3.1 Schematic illustration of device structure in this work.



Figure 3.2 Characteristic interface defect generation (CP data) under SHH injection and subsequent CVS.



Figure 3.3 Weibull distribution function versus injected-hothole-charge-to-breakdown (Q_{bd}^{hh}) under SHH stress.



Figure 3.4 Weibull distributions of Q_{bd} for devices under CVS with different amount of pre-injected hot holes by using SHH stress. The bars indicate the 95% confidence intervals for modal $Q_{bd}s$.

Chapter 4

Sigmoidal Defect Generation

4.1 Overview

The goal of this chapter is to investigate the defect generation as a function of injected charges. The defect generation rate (P_g), which is defined as the derivative of generated defects respect to the injected charges ($\Delta N_{it}/\Delta Q_{inj}$), during constant voltage stress (CVS) is investigated by using short-time voltage pulses over large fluence range. It is found that P_g is not constant during CVS and the voltage acceleration of P_g in the linear defect generation regime is similar to that of the reciprocal of Q_{bd}. Possible mechanisms that can cause the nonlinear defect generation behavior is discussed and the effect of the change of carrier capture cross (σ) during CVS to this nonlinear behavior was investigated. From this study it is conclusively shown that the change of P_g during CVS can not be explained by the change of σ . However, the results conclusively indicate that the linear region of defect generation must be used to extrapolate Q_{bd}.

4.2 Introduction

As mentioned in section 2.4, device lifetimes or t_{bd} are of great interests in both industry and reliability research [112-114]. However, since the oxide degradation is a gradual process, it usually takes years for the gate oxide to breakdown at normal device operating conditions. Therefore, to study the oxide degradation process within a limited time frame, the gate oxide is usually stressed at accelerated conditions (higher stress voltage or current). The critical reliability parameters such as t_{bd} and Q_{bd} are then extrapolated from the accelerated conditions to operating conditions. Although t_{bd} at operating conditions cannot be measured directly, it can be extracted by measuring P_g value. If one assumes a linear rate of defect generation, P_g , which is defined as $\Delta N_{it}/\Delta Q_{inj}$, is also equal to N_{bd}/Q_{bd} . Then Q_{bd} can be found as $Q_{bd}=N_{bd}/P_g$. Since N_{bd} is weakly dependent on the gate voltage (V_g) [112], Q_{bd} (and therefore t_{bd}) can be extrapolated if the voltage dependence of P_g is known [114].

However, since P_g is found not a constant during a stress [115], inconsistent extrapolations can be obtained using this method. Figure 4.1 shows the extrapolation of t_{bd} from the stress condition (high stress voltage) to the normal device operating condition (low stress voltage) assuming linear and nonlinear defect generation [114]. The results show that different extrapolation methods can cause several orders of magnitude variations in the predicted t_{bd} at normal operating condition. Therefore, understanding defect generation and its relationship to the oxide breakdown is important in the prediction of device lifetime. Moreover, it is also interesting to understand what mechanisms cause P_g value changes during a stress.

4.3 Determination of Defect Generation Rates

To understand how P_g value changes with Q_{inj} , short-time voltage pulses with constant amplitude have been used to stress devices and the defect generation over large fluence range is observed. The pulses used in this work are uni-polar square pulses with constant voltage amplitude between 5.2 volts and 6 volts, and the frequency is between 100 Hz and 100 kHz. The devices used are n-channel MOSFETs with channel area 30 um² and gate oxide thickness approximately 3.5 nm. Charge pumping (CP) [116, 117] and stress induced leakage current (SILC) [114, 118] measurements are performed to determine the average interface state density (D_{it}) and oxide bulk trap density (N_t), respectively. An additional 50- Ω resistor is connected in parallel to the gate probe during pulse stress to match the circuit impedance and maintain the square pulse waveform at the probe end as shown in Figure. 4.2.

From the CP theory introduced in section 2.4.3, it demonstrated that the peak CP current, $I_{CP,MAX}$, is linearly proportional to N_{it} . Since the defect generation rate is proportional to the number of existing defects at any moment [119, 120], the relative increase of interface state density ($\Delta N_{it}/N_{it0}$) instead of the absolute value of N_{it} is generally used in the study of defect generation mechanisms. Therefore, the relative change of $I_{CP,MAX}$ ($\Delta I_{CP,MAX}/I_{CP0,MAX}$) instead of $I_{CP,MAX}$ itself is plotted respect to Q_{inj} for comparison.

Figure 4.3 shows the relative increase in peak CP current ($\Delta I_{CP,MAX}/I_{CP0,MAX}$) as a function of Q_{inj} at different frequencies. It can be seen that the pulse frequency does not affect the defect generation rate, which is consistent with previous report [121]. The frequency independent defect generation means that this pulse stress technique can be used to obtain defect density at extremely small Q_{inj} . The linear defect generation curve is also shown in Figure 4.3 for comparison. It can be seen clearly that defect generation changes from a linear regime at low Q_{inj} (for $Q_{inj} \sim 0.1$ C/cm²) to a saturated regime at high Q_{inj} (for $Q_{inj} > 10$ C/cm²). The same result is also observed from SILC data (not shown).

The nonlinear defect generation curve has also been observed by other research groups and its effects on t_{bd} extrapolation at device operating voltage have been discussed [114]. In order to determine which P_g value can be used to extrapolate t_{bd} or Q_{bd} correctly, the voltage accelerations of P_g value in different regimes are compared. At first, a linear regression is used to fit the defect generation data, which is obtained by stressing a group of 10 devices at a certain stress voltage, as shown in Figure 4.4. This fitting curve represents the characteristic defect generation behavior at each stress condition and therefore, all fitting curves obtained from different stress voltages are compared, as shown in Figure 4.5. From this figure it can be seen that defect generation curves at different stress voltages all show the same saturated tendency at high Q_{inj} .

To investigate the change of voltage acceleration of P_g in different regimes in detail, the same fitting method used in Figure 4.5 is used to plot P_g as a function of Q_{inj} , as shown in Figure 4.6. In the linear regime, significant voltage acceleration of

 P_g is observed. However, this voltage acceleration becomes less significant in the saturation regime where all defect generation curves tend to converge. Moreover, it is also noticed that P_g in the "linear" regime is not constant either but changes relatively slower than in saturation regime.

Since P_g is not a constant during a stress, it is necessary to determine which P_g value in different regimes could be used for Q_{bd} or t_{bd} extrapolation. It is reminded that P_g is in proportional to the reciprocal of Q_{bd} by definition under the constant P_g assumption. Therefore, the voltage accelerations of P_g in the linear regime and in the saturation regime with the voltage acceleration of the reciprocal of Q_{bd} are compared, as shown in Figure 4.7. The P_g values compared in certain regime (linear or saturated) are chosen at the same Q_{inj} in different voltages. It can be seen that P_g in the linear regime has similar voltage acceleration as the reciprocal of Q_{bd} , but P_g in the saturation regime is less dependent on voltage. This result suggests that the P_g value in the linear regime can be better used to extrapolate Q_{bd} than that in the saturation regime. To conclusively confirm this point, further work is required on this issue and more experimental data is needed.

4.4 Discussion of Possible Mechanisms for the Nonlinear Behavior of Defect Generation

The non-linearity of defect generation has drawn attention from researchers and different theories have been proposed to explain it. Patrikar *et al.* suggested that the nonlinear defect generation is because the trapped electrons/holes are de-trapped and this trapping/de-trapping reaction will slow down the defect generation at high injection fluence [122]. Hu suggests that the nonlinear behavior is because hydrogen atom diffuses from the broken silicon bond, and therefore it should follow a power law behavior [123]. Sune *et al.* suggest that the non-linearity observed from CP data is because σ changes during stresses [124]. Some of other researchers think the nonlinearity is due to the inadequacy of the measuring techniques [112].

Figure 4.8 compares the characteristics of de-trapping model and hydrogen diffusion model with experimental data. The Hydrogen diffusion model requires that the power law exponent be between 0.5 and 1 but the experimental data shows that the exponent in the saturation regime is about 0.3. For the de-trapping model, it shows that the defect generation rate at high Q_{inj} is almost zero, which has never been observed from experiments. Therefore, neither of these two models can explain the defect generation over the whole fluence range. As for the effect of σ change on the observed nonlinear relative increases of peak CP current and SILC, it is suggested that since σ is found not a constant during stress, the relative increases of peak CP current and SILC are not really proportional to D_{it} and N_{ts} respectively. Therefore, the nonlinear relative increases of peak CP current and SILC may still lead to linear D_{it} and N_t generations due to this σ change effect [125]. In the following part of this section, this issue will be addressed. However, since it is difficult to extract σ from SILC data, in the following analysis σ will be extracted from CP data only and the behavior of σ inside bulk oxide is assumed to be similar to that at the interface.

By using the standard two-level charge pumping method with sinusoidal waveform as the pumping signal, both σ and D_{it} could be extracted at the same time [117, 125]. In this work, pulse stresses are interrupted periodically when the CP

measurement is performed and σ and D_{it} are extracted. Therefore, σ and D_{it} could be monitored as a function of injected charge. Figure 4.9 shows σ as a function of Q_{inj} . As it can be seen, σ does change during voltage stress and it decreases with Q_{inj} . However, the nonlinear generation behavior of D_{it} is still observed, as shown in Figure 4.10. It suggests that the change of σ can not explain the non-linearity of defect generation.

To show the effect of σ change on defect generation over Q_{inj} , the relative change in D_{it} assuming constant σ with that in which σ changes with Q_{inj} are compared, as shown in Figure 4.11. The linear defect generation curve is also shown in this figure for comparison. It is shown that the D_{it} generation with changing σ is closer to linear generation than the data with constant σ at low Q_{inj} . However, a similar power law saturated behavior of D_{it} is still observed at high Q_{inj} . This result suggests that the saturated behavior of defect generation is unlikely to be explained by the change of σ . It is also noticed from Figure 4.11 that the difference between D_{it} curves with constant σ and changing σ is small, except at very small injected fluence. Therefore, the general assumption that the relative increase of peak CP current is proportional to D_{it} generation is still valid and the conclusion obtained in section 4.3 still holds.

To get a better idea of how possible it is for the change of σ to be fully responsible for the saturation of defect generation at high Q_{inj}, the experimental CP data and the linear D_{it} generation curve are used to calculate the necessary corresponding σ over Q_{inj}. The result is shown in Figure 4.12. The experimental measured σ as shown in Figure 4.9 is re-plotted and also shown for comparison. It can be seen that in order to obtain linear D_{it} generation from the measured saturated CP data, σ has to increase instead of decrease as observed from experiment. Also, it has to increase to an unreasonable large value. This result suggests it is unlikely that the saturation of defect generation can be explained completely by the change of σ .

4.5 Conclusion

In this chapter, it has been shown that the voltage acceleration of P_g in the linear generation regime is closer to the voltage acceleration of the reciprocal of Q_{bd} than that in the saturation regime. This result suggests the P_g value in the linear regime can be better used for Q_{bd} and t_{bd} extrapolation to device normal operating conditions. It is also shown that the saturated behavior of defect generation can not be solely explained by the change of σ . Further work is necessary to determine the reason behind the fluence dependence of P_g .



Figure 4.1 Extrapolation of t_{bd} to device normal operating condition (low stress voltage) from accelerated stress condition (high stress voltage). It shows different extrapolations can cause the variations in predicting device lifetime (t_{bd}) as large as several orders of magnitude [114].



Figure 4.2 Illustration of experimental setup for the pulse stress experiment.



Figure 4.3 Defect generations at different frequencies under pulse stress. The overlaps of these curves showed that this technique can be used to measure defect density at a large range of fluence.



Figure 4.4 Illustration of linear regression fitting curve of defect generation at certain stress voltage.



Figure 4.5 Comparison of linear regression fitting curves of defect generations at different stress voltages.



 $\begin{array}{ll} Figure \ 4.6 & The \ comparison \ of \ linear \ regression \ fitting \\ curves \ of \ P_g \ at \ different \ stress \ voltages. \end{array}$



 $\begin{array}{ll} Figure \ 4.7 & Comparison \ of \ voltage \ accelerations \ of \ P_g \ in \ linear \\ regime, \ saturated \ regime \ and \ the \ reciprocal \ of \ Q_{inj}. \end{array}$


Figure 4.8 Comparison of defect generation of de-trapping model, hydrogen diffusion model and experimental data.



Figure 4.9 Extrapolated carrier capture cross section change over injected fluence.



Figure 4.10 Extracted averaged interface state density change over injected fluence.



Figure 4.11 Comparison of CP data curve (σ is assumed constant) and D_{it} generation curve (σ is changing with Q_{inj}) with linear defect generation curve.



Figure 4.12 Comparison of carrier capture cross sections over injected fluence: experimentally extrapolated (σ) and calculated from linear defect generation assumption (σ ').

Chapter 5

The Effect of Mobility Degradation to Threshold Voltage Shifts for Ultrathin Gate Oxides

5.1 Overview

In the previous chapter, the stress-induced defects as a function of injected charges are studied by using interrupted short-time pulse stress method and the nonlinear relationship is revealed. These generated defects affect device parameters such as gate leakage current density (J_g), carrier channel mobility (μ), and device threshold voltage (V_{th}) through various mechanisms. These parameters are used not only to monitor oxide degradation but also as the bases for building up empirical oxide degradation models. To ensure these models are correct, it is important and necessary to study the mechanisms that affect these parameters.

Therefore, the goal of this chapter is to investigate the dominant mechanism that causes device threshold voltage (V_{th}) shift. V_{th} shifts of p- and n-channel MOSFETs during a stress are analyzed from both experiment and simulation. The result of the analysis showed that V_{th} shift is mainly induced by the carrier channel mobility degradation. This result can explain the polarity dependence of V_{th} shifts in p- and n-channel MOSFETs. Besides, it suggested that the commonly accepted idea that V_{th} shifts are due to Coulombic charge generation in the oxide affecting the surface potential is not accurate. It also suggested that proposed oxide degradation mechanisms based on V_{th} shifts measured using I_d - V_g may not be accurate.

5.2 Introduction

The degradation of the gate oxide of metal-oxide-semiconductor field effect transistors (MOSFETs) during an electrical stress is an extensive issue in device reliability and has been investigated from various aspects for many years [126-128]. The proposed empirical models for the oxide degradation mechanisms based on experimental observations are monitored by using various techniques such as charge pumping (CP) [129, 130], stressed induced leakage current (SILC) [131-132], DCIV [133], gate leakage current [134-136], and I_d-V_g [132, 137-139], etc. The device V_{th} shift measured by using MOSFET I_d-V_g characteristics is commonly assumed to be due to shifting of the characteristics by Coulombic charge build-up ($\Delta Q/C_{ox}$) in the dielectric [137-139]. However, the contribution of the carrier channel mobility degradation to V_{th} shift is not well characterized. Therefore, to distinguish the

contributions of Coulombic charge effect and the carrier channel mobility degradation to V_{th} shift is the main focus of this work.

5.3 V_{th} and V_{FB} Shift Measurements

The devices under test were fully processed p- and n-channel MOSFETs with an area 50x50 μ m² and gate oxide thickness of 2 nm. Constant voltage stress (CVS) was applied at room temperature, and V_{th} was measured periodically during the stress. In order to explore the mechanisms causing V_{th} shifts, V_{th} shift measured using the I_d-V_g method is compared with flat band voltage (V_{FB}) shift measured by Capacitance-Voltage (C-V) method. For devices which V_{FB} shifts were measured and calculated from the C-V method using a LRC meter, they were also stressed by using the same instrument to reduce noise. Accordingly, both p- and n-channel devices were divided into two groups in this work. In the first group, CVS was applied by using a HP4156B semiconductor analyzer at ± 3.7 V, and V_{th} was measured by using the I_d-V_g SPICE method [132, 136]. In the second group, CVS was applied by using a HP4248A LCR meter with DC bias at ± 3.7 V, and the oscillation voltage amplitude was 5 mV at 10 Hz. VFB shifts of devices in this group were obtained by first measuring the capacitance change (ΔC) at a chosen measurement voltage (V_{measured}) and then being divided by the derivative of the initial C-V curve (dC/dV) at V_{measured}, as shown in Figure 5.1. The frequency and oscillation voltage amplitude for V_{FB} measurement were 100 kHz and 50 mV, respectively.

Figure 5.2 shows the comparison of V_{th} shifts using I_d - V_g method and V_{FB} shifts using C-V methods in n-channel MOSFETs stressed in inversion condition.

Each measurement was repeated seven times after each stress and three different devices were tested. The variances between the maximum and minimum experimental data values were smaller than the plotted symbols. It can be seen clearly that V_{th} shift measured by the I_d - V_g method is much larger than V_{FB} measured by the C-V method. Figure 5.3 shows the measurement results for p-channel MOSFETs and similar conclusions are obtained. Although it is not shown, it should be noticed that the general behaviors of measured V_{th} and V_{FB} shifts for both channels of MOSFET stressed in accumulation condition are similar to those observed in inversion condition.

It is believed that the V_{th} shift measured by the I_d - V_g method can be induced by Coulombic charge in the oxide affecting the surface potential or the degradation of the carrier channel mobility causing a decrease in the drain current at a given gate bias. On the other hand, the measured V_{FB} shift is believed to be completely due to the Coulombic charge accumulation inside the oxide. Since the measured V_{th} shift from I_d - V_g method is much larger than V_{FB} shift from C-V method, it suggests that the mobility degradation during CVS should have a larger effect than the Coulombic charge accumulation inside the oxide on V_{th} shift measured by I_d - V_g method.

5.4 Simulation of $V_{th}\,Shift$

To verify the hypothesis and determine the dominating mechanism causing V_{th} shifts, the North Carolina State University (NCSU) Mod2D mobility extraction program was used to simulate V_{th} shifts. In this program, the interface scatter density (N_{scat}) is used to model Coulombic scattering (and, hence, mobility reduction). The

fixed oxide charge density (Q_{ox}) is used to model the Coulombic charge in the oxide affecting the surface potential. To distinguish the contributions from N_{scat} (mobility degradation effect) and Q_{ox} (Coulombic shift effect) on V_{th} shift, three different scenarios that cause V_{th} shift were simulated: 1) N_{scat} and Q_{ox} increasing at the same rate, 2) N_{scat} increasing at a certain rate with Q_{ox} constant, and 3) Q_{ox} increasing at a certain rate with N_{scat} constant. The initial N_{scat} and Q_{ox} are assumed to be the same and equal to each other in all cases. The simulation results for n- and p- channel MOSFETs are shown in Figures 4 and 5, respectively.

The simulation results clearly show that V_{th} shift due to N_{scat} is much higher than Q_{ox} in both n- and p-channel MOSFETs. It suggests that the carrier channel mobility degradation is the main mechanism causing V_{th} shifts in both types of MOSFET. This simulation result can explain the polarity dependence of V_{th} shifts in n- and p- channel MOSFETs without assuming charges of opposite polarities are generated at the same fabricated oxide in both types of MOSFET. Since the carrier channel mobility degraded during a stress [140, 141], it requires more positive (negative) gate voltage for the n-channel (p-channel) MOSFET to keep up the drain current. As a result, the absolute values of V_{th} in both types of MOSFET are increased, which automatically induces the polarity dependence.

It should be reminded that the simulation results are based on the assumption that the initial N_{scat} and Q_{ox} are equal to each other. However, since the effective N_{scat} and Q_{ox} at Si/SiO₂ interface are affected by the physical location and distribution of defects, the effective N_{scat} and Q_{ox} at interface may not be the same. Therefore, their contributions to V_{th} shifts may change. Nevertheless, the results of this simulation suggest that the commonly accepted idea that V_{th} shifts are due to Coulombic charge generation in the oxide affecting the surface potential may not be accurate. It also suggests that proposed oxide degradation mechanisms based on V_{th} shifts measured using the I_d-V_g method may not be accurate.

5.5 Conclusion

In this work, it is shown by using modeling that the dominant cause of V_{th} shift measured using I_d - V_g is not shifting of the characteristic by Coulombic charges, but is instead due to modification of the I_d - V_g characteristic by mobility degradation if the effective quantity of N_{scat} and Q_{ox} at interface are the same. This result can explain the polarity dependence of V_{th} shifts in p- and n-channel MOSFETs. Moreover, it suggests that oxide degradation mechanisms based on V_{th} shifts measured using I_d - V_g may not be accurate. Finally, it has important consequences on interpreting data and developing an understanding of dielectric reliability physics.





Figure 5.2 Comparison of V_{th} shifts measured by I_d - V_g and V_{FB} shifts measured by C-V methods for n-type MOSFET stressed in inversion





Figure 5.4 Simulation results of V_{th} shifts in n-type MOSFET at different trap generation conditions.



Figure 5.5 Simulation results of V_{th} shifts in p-type MOSFET at different trap generation conditions.

Chapter 6

Electrical Characterization of Spatial Distributions of Trapping Centers in HfO₂/SiO₂ Stacked Dielectrics

6.1 Overview

As mentioned in Chapter 1, high-k dielectrics will be used to replace SiO_2 as the gate material for future MOS devices to reduce leakage current. It is known that high-k dielectrics have significant amount of intrinsic defects (trapping centers) and these defects are distributed not only at the interface but also inside the bulk high-k dielectrics. To correctly characterize these defects would be essentially helpful to understand the properties of these traps.

Therefore, the goal of this chapter is to develop a methodology to characterize the spatial distribution of these traps. The methodology is based on charge pumping (CP) measurement and is used to extract the spatial profile of traps inside SiO₂/HfO₂ stacked dielectrics. From simulation results it showed that different parts of the total traps inside high-k dielectrics would be probed during CP measurement by changing measurement parameters. Traps at different locations inside stacked dielectrics are therefore characterized separately and their spatial profile is revealed. From the spatial profiles of traps, SiO₂ region, SiO₂/HfO₂ diffusion region and HfO₂ region in the stacked dielectric are clearly identified. The correlation between the shift of SiO₂/HfO₂ diffusion region and the difference of the interfacial layer thickness of stacked dielectrics demonstrate this methodology is accurate and reliable.

6.2 Introduction

As the gate oxide thickness of the Metal-Oxide-Semiconductor Field-Effect-Transistor (MOSFET) is scaled down, the gate leakage current increases exponentially raising concerns regarding various reliability issues of the gate oxide [142-150]. To reduce the large gate leakage while further scaling equivalent oxide thickness, substitute materials with high dielectric constant (high-k), such as Al₂O₃, HfO₂ and ZrO₂, have been proposed to replace SiO₂ [151-159]. Among these materials, HfO₂ has been considered as one of the most promising substitute materials due to its thermodynamic stability on silicon, its large dielectric constant (25) and reasonable bandgap (~5.7 eV) [153,157-161]. One major difference between high-k dielectrics and SiO₂ is the large amount of traps in the high-k. These traps are believed to be distributed not only at the interface but also inside the bulk dielectric [162, 163]. However, the exact spatial distribution of these traps is still not clear. Knowing the spatial distribution of the traps is an important part of understanding of the physical nature of the defects and how they affect device properties. The main purpose of this work is to develop a CP method to characterize the spatial distribution of the traps in high-k dielectrics.

6.3 Simulation of the Probable Range in CP Measurement

CP has been used to study interface and near-interface traps in the Si/SiO₂ system for more than thirty years and has been used widely [164-166]. When performing CP measurement, the source and drain are usually grounded or applied with a small reverse bias. Periodic pulses are applied to the gate and drive the channel region into inversion and accumulation conditions periodically [167]. A dc CP current measured from substrate is attributed to the recombination of trapped electrons and holes. If the pulse amplitude (V_a) is kept constant and the pulse base voltage (V_{base}) is swept from flat band voltage (V_{FB}) to threshold voltage (V_{th}) or vice versa, a maximum dc CP current ($I_{CP,MAX}$) will be observed. Conventionally, $I_{CP,MAX}$ is expressed as [168],

$$I_{CP,MAX} = q N_{it} f A_G....(6.1)$$

where q (coul) is the unit Coulombic charge, f (s⁻¹) is the frequency of the applied pulses, A_G is the effective channel area (cm⁻²) and N_{it} (cm⁻²) is the total number of interface traps per area. In this expression, N_{it} is typically considered to be independent of the distance of traps from the interface. This assumption is valid in Si/SiO₂ system for which most traps are located close to the Si/SiO₂ interface and are

probed by the CP measurement at moderate frequencies [169, 170]. However, in high-k dielectrics since traps are not only at the interface but also inside the bulk dielectric, traps located away from the interface may not be probed during CP measurement. To include the effect of the spatial distributions of traps to $I_{CP,MAX}$, equation (6.1) is re-written as

$$I_{CP,MAX} = q \ f \ A_G \ N_{mit}$$

= $q \ f \ A_G \int_{0}^{x_d} \int_{E_{min}}^{E_{max}} N_{it}(x) \Delta F(x, E_t) dE_t dx.....(6.2)$

where N_{mit} is the measured N_{it} during CP measurement and could be expressed as the double integral of the multiplication of N_{it} and an additional term ΔF . ΔF indicates the probability that a trap can be probed by CP measurement and is a function of the distance from the Si-substrate/gate-dielectric interface (x) and the trap energy (E_t). The upper and lower limits of the double integral are the energy range ($E_{max} - E_{min}$) and the maximum probable depth (x_d) in which traps inside can be probed. It will be shown in the following that ΔF is strongly affected by the parameters of the pulses applied to the gate during CP measurement such as V_a , pulse on/off time (t_{on}/t_{off}), and pulse rise/fall time (t_r/t_f). From equation (6.2) it can also be seen that N_{mit} and $I_{CP,MAX}$ are functions of ΔF and therefore will be affected by the pulse parameters as well. It suggests that different $I_{CP,MAX}$ may be obtained by using different pulse parameters due to different parts of traps been characterized. It is therefore, crucial to understand how ΔF and $I_{CP,MAX}$ are affected by these pulse parameters during CP measurement.

The explicit expression of ΔF can be obtained by considering the capture and emission processes of electrons and holes at traps described in Shockley-ReadHall theory [171, 172]. From Shockley-Read-Hall theory, four different possible emission and capture processes can happen to a trapping center at energy E_t : electron capture, electron mission, hole capture, and hole emission. The corresponding rates for these processes are the electron capture rate (c_n), electron emission rate (e_n), hole capture rate (c_p) and hole emission rate (e_p). The overall rate equation can be express as

$$\frac{dF}{dt} = (c_n + e_p) - (c_n + c_p + e_n + e_p) \cdot F....(6.3)$$

where F is the occupancy function and indicates the probability of a trapping center being occupied by an electron (or hole) at any given time t.

To understand how F changes during CP, it is assumed that trapezoidal pulses are applied to the gate as a function of time and the corresponding F values can be found. However, it is noticed that c_n and c_p are functions of the Si substrate Fermi level (E_f) and therefore are functions of time during t_r and t_f [173]. To simplify the problem and obtain an analytic expression of F, both t_r and t_f are assumed to be negligible, which means the trapezoidal pulses are approximated by square pulses. Under this approximation, c_n and c_p are constant and equation (6.3) is a first-order ordinary differential equation. The solution to this equation is

where t is the time and t = 0 is defined as the points at which E_f reaches the quasisteady state condition. Figure 6.1 illustrates F as a function of time for a square pulse. It can be seen that during t_{on} , F increases from its minimum value to its maximum value; while during t_{off} , F decreases from its maximum value to its minimum value. Since t in equation (6.4) is the time after E_f reaches quasi-steady state, its origin can be assigned at any edges of the square pulse, as shown in Figure 6.1. The maximum and minimum F values can then be expressed as:

$$F_{\max} = F_{\min} \cdot e^{-(c_{n,on} + c_{p,on} + e_n + e_p) \cdot t_{on}} + \frac{(c_{n,on} + e_p)}{(c_{n,on} + c_{p,on} + e_n + e_p)} (1 - e^{-(c_{n,on} + c_{p,on} + e_n + e_p) \cdot t_{on}}) \dots (6.5)$$

$$F_{\min} = F_{\max} \cdot e^{-(c_{n,off} + c_{p,off} + e_n + e_p) \cdot t_{off}} + \frac{(c_{n,off} + e_p)}{(c_{n,off} + c_{p,off} + e_n + e_p)} (1 - e^{-(c_{n,off} + c_{p,off} + e_n + e_p) \cdot t_{off}}) \dots (6.6)$$

where $c_{n,on}$ and $c_{n,off}$ are the electron capture rates during t_{on} and t_{off} and $c_{p,on}$ and $c_{p,off}$ are the hole capture rates during t_{on} and t_{off} .

 F_{max} and F_{min} indicate the probabilities that a trapping center is occupied by electrons at the end of t_{on} , and t_{off} , respectively. The difference between F_{max} and F_{min} $(\Delta F = F_{max} - F_{min})$ indicates the probability of a trapping center being occupied by an electron and a hole alternatively (electron-hole recombination) during a complete pulse period. In other words, it indicates the probability a trapping center can be probed during CP measurement and contribute to $I_{CP,MAX}$. By using equations (6.5) and (6.6), ΔF can be expressed as

$$\Delta F = F_{\rm max} - F_{\rm min}$$

$$=\frac{(e^{(c_{n,on}+c_{p,on}+e_{n}+e_{p})t_{on}}-1)(e^{(c_{n,off}+c_{p,off}+e_{n}+e_{p})t_{off}}-1)(-c_{n,off}(c_{p,on}+e_{n})+c_{n,on}(c_{p,off}+e_{n})+(-c_{p,on}+c_{p,off})e_{p})}{(e^{(c_{n,on}+c_{p,on}+e_{p})t_{on}+(c_{n,off}+c_{p,off}+e_{n}+e_{p})t_{off}}-1)(c_{p,on}+e_{n}+c_{n,on}+e_{p})(c_{p,off}+e_{n}+c_{n,off}+e_{p})}..(6.7)$$

In the derivation of ΔF , the input trapezoidal pulses are approximated by square pulses and assumed to have zero t_r and t_f . It implies that E_f changes abruptly from the Fermi level at accumulation condition ($E_{f,acc}$) to the Fermi level at inversion condition ($E_{f,inv}$) when the pulse voltage level (V_{top} or V_{base}) switches. Therefore, the time periods for electrons and holes to occupy traps are exactly t_{on} and t_{off} respectively. As for non-zero t_r and t_f , since now E_f changes gradually during t_r and t_f , the surface electron (hole) density increases exponentially during t_r (t_f). Therefore, the "effective" time period for electrons (holes) to occupy traps will be longer than t_{on} (t_{off}). This effect of t_r and t_f on the trap occupancy by electrons and holes will be considered later.

Non-zero t_r and t_f have other effects on ΔF . Ideally, all traps with energy E_t between $E_{f,inv}$ and $E_{f,acc}$ should be able to be probed during CP since E_f is pinned at these two energy levels during t_{on} and t_{off} . However for non-zero t_r (t_f), trapped holes (electrons) with energies close to $E_{f,acc}$ ($E_{f,inv}$) can be emitted to the substrate (source and drain) instead of being recombined. The effective minimum and maximum trap energy between which recombination occurs are then given by equations (6.8) and (6.9) [168]:

$$E_{em,h} = E_i + k T \ln(v_{th} \sigma_p n_i \frac{|V_{FB} - V_{th}|}{|V_a|} t_r + e^{\frac{(E_F, acc - E_i)}{kT}})......(6.8)$$

$$E_{em,e} = E_i - k T \ln(v_{th} \sigma_n n_i \frac{|V_{FB} - V_{th}|}{|V_a|} t_f + e^{\frac{(E_i - E_F, inv)}{kT}})....(6.9)$$

where $E_{em,h}$ ($E_{em,e}$) is the effective maximum (minimum) trap energy level, E_i (n_i) is the intrinsic Fermi level (carrier density), v_{th} is the thermal velocity and σ_n (σ_p) is the electron (hole) capture cross section.

On the other hand, the effective carrier capture cross sections are known as functions of x and can be expressed as [169, 174]

$$\sigma_{n/p}(x) = \sigma_{n/p}(0)e^{-\frac{x}{\lambda_{n/p}}}$$
.....(6.10)

where $\sigma_{n/p}(x)$ is the electron/hole capture cross section at any depth x, while $\sigma_{n/p}(0)$ is the electron/hole capture cross section at the interface and $\lambda_{n/p}$ is the characteristic tunneling distance of electron/hole. By using equation (6.10), equations (6.8) and (6.9) are re-written as

$$E_{em,e} = E_i - k T \ln(v_{th} \sigma_n(0) e^{-\frac{x}{\lambda_n}} n_i \frac{|V_{FB} - V_{th}|}{|\Delta V_A|} t_f + e^{\frac{(E_i - E_F, inv)}{kT}})$$

$$\approx E_i + k T \ln(v_{th} \sigma_n(0) n_i \frac{|V_{FB} - V_{th}|}{|\Delta V_A|} t_f) - \frac{kT}{\lambda_n} x....(6.12)$$

By combining equations (6.7), (6.11) and (6.12), a 3-D ΔF simulation contour as a function of E_t and x can be plotted as shown in Figure 6.2. Traps located inside the trapezoidal plateau with ΔF equal to one have the maximum probability to be probed and contribute to $I_{CP,MAX}$ during CP measurement. It can be seen that the detectable trap energy range is narrower at the interface and is expanding with x, as described in equations (6.11) and (6.12). It can also be seen that ΔF drops from one to zero at about 8 Å, and it suggests that all traps beyond this depth can not be detected during CP measurement. Therefore, a probable depth during CP measurement can be defined accordingly. It should be reminded that the contour shown in Figure 6.2 depends on applied pulse parameters during CP measurement. Therefore, by changing the pulse parameters different contours can be obtained, as shown in Figure 6.3. Figure 6.3 (a) and (b) show the 2-D simulation contours of ΔF equals to 0.5 with different pulse parameters. In both contours t_{on} is set equal to t_{off} and is equal to 50 ns in (a) while is equal to 1 µs in (b), and the other pulse parameters are kept the same. It can be seen that at lower frequency (longer t_{on} and t_{off}) the probable depth is deeper which suggests that electrons and holes have longer time to penetrate into bulk dielectric to occupy traps at deeper depth and contribute to I_{CP,MAX}.

As mentioned earlier, the effect of non-zero t_r and t_f on the derivation of equation (6.7) is to increase the effective t_{on} and t_{off} , and effectively increases the probing depth. However, since the real probing depth is affected strongly by dielectric parameters (which will be shown in the following), this effect on the increase of effective probing depth is not crucial and the ignorance of this effect is valid without affecting the further semi-quantitive analysis.

6.4 Experiment

The devices used in this work are fully processed MOSFETs with HfO_2/SiO_2 stacked gate dielectrics. High-k gate dielectric transistors were fabricated on 200mm p/p+ epitaxial Si <100> wafers using a standard CMOS process with 1000 °C/10 sec

dopant activation anneal. The gate stacks were formed by depositing a 3 nm ALD HfO_2 dielectric on various scaled thermal oxide interface layers (IL) created by the controlled etch-back of a 1.9 nm thermal oxide. The high-k film deposition was followed by a 700 °C anneal in NH₃ ambient, after which, a gate electrode was formed by CVD TiN with poly-Si cap [175].

CP measurement is performed by applying periodical trapezoidal pulses with fixed t_r/t_f and V_a generated by HP8112A pulse generator to the gate. The electronhole combination dc current is measured from the substrate by using HP4156B semiconductor analyzer. To probe traps at different depth in the dielectric, t_{on} and t_{off} of the applied pulses are kept the same and change from 50 ns to 100 ms. Figure 6.4 shows $I_{CP,MAX}$ as a function of $t_{on/off}$. It can be seen that $I_{CP,MAX}$ decreases with increasing t_{on}/t_{off} and then saturates at long $t_{on/off}$. This saturation indicates that $I_{CP,MAX}$ is dominated by the gate leakage current instead of the electron-hole recombination current. This leakage current limits the maximum probable depth in dielectrics while performing CP measurement. Meanwhile, the measured $I_{CP,MAX}$ at all t_{on}/t_{off} is corrected by using this leakage current to get the $I_{CP,MAX}$ due to electron-hole combination.

6.5 Results and Discussion

One issue associated with this methodology is to get the probing depth from t_{on}/t_{off} . The conversion between t_{on}/t_{off} and x is affected by theoretical values of dielectric parameters such as effective electron/hole mass inside dielectrics (m_{e/h}), effective electron/hole barrier height ($\Phi_{e/h}$), $\sigma_{n/p}$ (0), and V_a [169, 170]. This effect is

shown in Figure 6.5. Figure 6.5 shows that N_{mit} is plotted as a function of x by using two different sets of parameters. For the open circle, the used parameters are $m_{e/h}$ = 0.5/0.4 eV, $\Phi_{e/h} = 3.1/3.8$ eV, $\sigma_{n/p} (0) = 10^{-14}/10^{-16}$ cm⁻², which are the commonly accepted values for pure SiO₂ [176, 177]. For the closed circle, the used parameters are $m_{e/h} = 0.1/0.1$ eV, $\Phi_{e/h} = 1.3/3.3$ eV, $\sigma_{n/p} (0) = 10^{-14}/10^{-15}$ cm⁻², which are the estimated values for pure HfO₂. These values are estimated so that the location of the plotted curve is consistent with the thickness of SiO₂ interfacial layer from fabrication. In both cases V_a is equal to 1.2 V. It can be seen that by using different sets of parameters, the curve is not only shifted but also stretched out along x-axis. Since the device under test has a SiO₂/HfO₂ stacked gate dielectric, it is expected that the real values of these parameters should be somewhere in between the chosen two set values and they should also change with the probing depth. Therefore, the maximum detectable depth range is from end to end of these two curves which is about 1.6 nm as shown in the figure. However, since the accurate values of these parameters are not clear, the depth shown in the x-axis can only be a reference and only semi-quantitive analysis can be provided. Meanwhile, to simplify the analysis and make it easier to compare experimental results from devices with different interfacial layer thickness, parameters with values for HfO₂ will be used for HfO₂/SiO₂ stacked dielectrics in the following analysis.

Figure 6.6 shows N_{mit} at different x for devices with interfacial layer oxide thickness ranging from 1nm to 2 nm. N_{mit} from pure SiO₂ dielectric is also shown for comparison. It should be reminded that N_{mit} (x) is an accumulation function which counts the total number of traps per area from Si/SiO₂ interface to depth x. With that it can be seen that the values of N_{mit} at the smallest x for all curves are very close, which suggests that all dielectrics have approximately the same amount of traps within a shallow depth. These traps are attributed to the commonly observed interface traps at Si/SiO₂ interface. For SiO₂ dielectric, its curve is flat and keep N_{mit} a constant through out the detectable depth range. It suggests no further traps exist inside the bulk SiO₂ which is expected and is consistent with observations from other groups [169, 170]. As for the curves for HfO₂/SiO₂ stacked dielectrics, N_{mit} increases with depth and the value is higher for the dielectric with thinner SiO₂ interfacial layer at the same depth. It suggests that additional traps are generated inside the bulk SiO₂ due to the diffusion phenomena at HfO₂/SiO₂ interface. For the dielectric with thinner interfacial layer, since HfO₂/SiO₂ interface is closer to Si/SiO₂ interface this diffusion is more prominent so that N_{mit} is higher. As for the curve for the dielectric with the thickest interfacial layer (2nm), it shows similar behavior as that observed from SiO₂ dielectric. It suggests that since the interfacial layer is thick, additional traps inside SiO_2 due to the diffusion at HfO_2/SiO_2 are still beyond the probable depth. Therefore, no additional prominent taps are observed in this dielectric.

To get the trap volume density, N_t (cm⁻³), the derivative of $N_{mit}(x)$ respect to x is taken and the result is shown in Figure 6.7. Since the dielectric with 2 nm SiO₂ interfacial layer does not show prominent additional traps, only dielectrics with thinner SiO₂ interfacial layers are plotted. It can be seen clearly that both devices show that N_t is low at shallow depth, then increases with x and finally tends to saturate. The result suggests that the probing region changes from pure SiO₂ region, passes SiO₂/HfO₂ diffusion region and then reaches pure HfO₂ region. The saturation

of N_t in the HfO₂ region may suggest that the trap distribution is relatively uniform. However, this result needs further work to be confirmed. Meanwhile, it can also be seen that the SiO₂/HfO₂ diffusion regions in these two dielectrics are shifted about 4 Å, which is close to the difference between the SiO₂ interfacial layer thickness of these two dielectrics. It suggests that although the accurate probing depth is unknown, the relative trap spatial distribution is accurate.

6.6 Conclusion

From the simulation result it has been shown that N_{mit} during CP characterization is not equal to N_{it} in high-k dielectrics. It is also shown that the probable range of traps in the dielectrics is affected by pulse parameters. The results are essentially important while comparing experimental results from different electrical characterization techniques. The results are also helpful to understand which portions of traps been probed while study the properties of traps in high-k dielectrics. By using this methodology, trap spatial profiles in the SiO₂/HfO₂ stacked dielectrics with different SiO₂ interfacial layer thickness are also shown in this report. The results clearly showed the change of N_t from SiO₂ layer to SiO₂/HfO₂ diffusion region and reaches HfO₂ layer. Although the accurate depth of traps are not clear due to the undetermined theoretical dielectric parameters, the relative shift of the trap profile is consistent with the difference of the interfacial layer thickness in different SiO₂/HfO₂ stacked dielectrics. It suggests that this methodology is accurate and reliable.



Figure 6.1. Illustration of the change of F value with respect to a square pulse.



Figure 6.2. 3-D ΔF contour simulation result. ΔF is equal to one within the trapezoidal plateau that indicates the region having the maximum probability been probed.





Figure 6.3. Simulation results of pulse parameters dependence of ΔF contour. In (a) both t_{on} and t_{off} are equal to 50 ns, while in (b) both t_{on} and t_{off} are equal to 1 μ s. All the other pulse parameters are the same in both diagrams.



Figure 6.4. I_{CP,MAX} is proportional to frequency. When frequency decreases, I_{CP,MAX} tends to saturate. This saturation indicates I_{CP,MAX} is dominated by gate leakage current instead of electron-hole combination current



Figure 6.5. The probing depth calculated in this methodology is strongly affected by dielectric theoretical values. For the open circle, the used parameters are $m_{e/h} = 0.5/0.4 \text{ eV}$, $\Phi_{e/h} = 3.1/3.8 \text{ eV}$, $\sigma_{n/p} (0) = 10^{-14}/10^{-16} \text{ cm}^{-2}$, which are the commonly accepted values for pure SiO₂. For the closed circle, the used parameters are $m_{e/h} = 0.1/0.1 \text{ eV}$, $\Phi_{e/h} = 1.3/3.3 \text{ eV}$, $\sigma_{n/p} (0) = 10^{-14}/10^{-15} \text{ cm}^{-2}$, which are the estimated values for pure HfO₂. In both cases V_a is equal to 1.2 V



Figure 6.6. The comparison of N_{mit} in dielectrics with different thickness of SiO₂ interfacial layer. It can be seen that at the same depth N_{mit} is higher for the dielectric having thicker interfacial layer. As for SiO₂ dielectric, N_{mit} is constant which indicates no further traps exist in bulk dielectric



Figure 6.7. The comparison of N_t in dielectrics with different thickness of SiO₂ interfacial layer. The pure SiO₂, SiO₂/HfO₂ diffusion and pure HfO₂ regions can be identified clearly in both dielectrics
Chapter 7

Summary and Future Work

7.1 Overview

Electrical characterizations on both ultra-thin SiO₂ and SiO₂/HfO₂ stacked dielectrics from various aspects are performed in this research work. The goal of this chapter is summarize important results from previous chapters in this dissertation. Possible further extended work will also be discussed in this chapter.

7.2 Summary

In this research work, the generation of electrical active defects in SiO_2 dielectrics during stresses has been characterized from different aspects. It is found that different electrical defects may be generated during CVS and SHH injection. This conclusion is supported by the observation that the Weibull slope for hot-hole-to-breakdown is much larger than that for Q_{bd} during CVS. It is also supported by the other observation that pre-injected hot holes do not contribute to dielectric degradation during the subsequent CVS.

It is also shown in this work that critical reliability parameters such t_{bd} can be extrapolated from accelerated test condition to device normal operating condition by using P_g value. However, since P_g is not a constant during stress, this extrapolation should be done by using P_g value in the linear generation regime. Although the real mechanism causing the non-constant P_g is still not well understood, it is shown that the change of defect capture cross section can not explain this phenomenon.

Meanwhile, the dominant mechanism that causes V_{th} shifts in both n- and pchannel MOSFETs are also investigated in this work. From both experiment and simulation results, it is found that channel mobility degradation is the main mechanism that causes this shift. Therefore, commonly accepted idea that V_{th} shift is due to the Coulombic charge accumulation at the interface is not accurate. This result also indicates that oxide degradation model based on Coulombic charge accumulation may not be accurate.

The electrical characterization on high-k dielectrics in this work was focus on the properties of the initial traps in high-k dielectrics. A methodology based on 2level CP measurement with various frequencies was used to revel the spatial profile of these traps. From measurement results, the spatial profile of traps in the SiO₂ region, SiO₂/HfO₂ diffusion region and HfO₂ region in the stacked dielectric are clearly identified. Although the exact probing depth in this methodology may not be accurate due to the unknown theoretical dielectric parameters, the correlation between the shift of SiO₂/HfO₂ diffusion region and the difference of the interfacial layer thickness of stacked dielectrics demonstrate this methodology is accurate and reliable.

7.3 Future work

Electrical defect generation in SiO₂ dielectrics has been studied extensively in the past a few decades. Although there is still no conclusive mechanism to describe the degradation process during electrical stresses, it is generally accepted that the generation of energetic carriers (holes or hydrogen species) are the dominant factor. Therefore, continuously focus on how these two species are generated and contribute to oxide degradation during various stress conditions such as CVS, SHH injection and negative bias temperature instability (NBTI) stress will be helpful to construct the whole picture of the physics behind the gate oxide degradation.

Meanwhile, the understanding of oxide degradation process in SiO_2 dielectric will also provide helpful information in studying the defect generation mechanisms in high-k dielectrics. Although there are sill many unknowns about high-k dielectrics, there is no doubt that high-k dielectrics such as HfO₂ will be used to replace SiO₂ in MOS devices in the future based on the need from industry. Therefore, it is expected that more research effort will be put into the filed of studying high-k dielectrics in the future. The future topics in high-k dielectrics shall include not only characterizing and improving the quality of high-k dielectrics but also understanding the defect generation in high-k dielectrics during electric stresses. It would be interesting and crucial to understand how high-k dielectrics degrade during electrics and how it is compared to SiO₂ dielectric. By comparing the characterization results from SiO₂ and high-k dielectrics, it can provide more information of electrical defect generation in dielectrics.

Reference

- W. Shockley and G.L. Pearson, "Modulation of Conductance of Thin Films of Semiconductors by Surface Charges," *Phys. Rev.*, vol. 74, pp. 232, 1948.
- J.R. Ligenza and W.G. Spizer, "Effect of Crystal Orientation on Oxidation Rates of silicon in High Pressure steam," *J. Phys. Chem.*, vol. 65, pp. 2011, 1961.
- [3] G. Moore, "Cramming More Component onto Integrated Circuits," *Electron.*, vol. 38, number 8, April 19, 1965.
- [4] www.intel.com/research/silicon/mooreslaw.htm.
- [5] E.M. Vogel, W.K. Henson, C.A. Richter, and J.S. Suehle, "Limitations of Conductance to the Measurement of the Interface State Density of Mos Capacitors with Tunneling Gate Dielectrics," *IEEE Trans. Electron Devices*, vol. 47, pp. 601, 2000.
- [6] J.S. Suehle, "Ultrathin Gate Oxide Reliability: Physical Models, Statistics, and Characterization," *IEEE Trans. Electron Devices*, vol. 49, pp. 958, 2002.
- [7] International Technology Roadmap for Semiconductors, 2004.
- [8] D.A. Buchanan and S–H. Lo, "Growth, Characterization and the Limits of Ultrathin SiO₂-Based Dielectrics for Future CMOS Applications," in *The Physics and Chemistry of SiO₂ and theSi-SiO₂ Interface, Proc. Vol. 96-1.* Pennington, NJ: Electrochem. Soc., pp. 3, 1996.

- [9] G. Timp et al., "Low Leakage, Ultra-thin Gate Oxides for Extremely High Performance Sub-100 nm nMOSFETs," in *IEDM Tech. Dig.*, pp. 930, 1998.
- [10] J.H. Stathis and D.J. DiMaria, "Reliability Projection for Ultra-thin Oxides at Low Voltage," in *IEDM Tech. Dig*, pp. 167, 1998.
- [11] Y. Taur and E.J. Nowak, "CMOS Devices below 0.1 um: How High Will Performance Go," in *IEDM Tech. Dig.*, pp. 215, 1997.
- [12] S.H. Lo, D.A. Buchanan, Y. Taur, and W. Wang, "Quantum-mechanical Modeling of Electron Tunneling Current from the Inversion Layer of Ultrathin-oxide nMOSFETs," *IEEE Electron Device Lett.*, vol. 18, pp. 209, 1997.
- [13] T.H. Ning, "Silicon Technology Directions in the New Millennium," in *Proc. Int. Reliability Physics Symp.*, pp. 1, 2000.
- [14] G-W. Lee, J-H. Lee, H-W. Lee, M-K Park, D-G. Kang, and H-K Youn, "Trap Evaluations of Metal/Oxide/Silicon Field-effect Transistors with High-k Gate Dielectric Using Charge Pumping Method," *Appl. Phys. Lett.*, vol. 81, pp. 2050, 2002.
- [15] S. Zafar, A. Callegari, E. Gusev, M.V. Fischetti, "Charge Trapping in High-k Gate Dielectric Stacks," in *IEDM Tech. Dig.*, pp. 517, 2002.
- [16] P. Masson, J-L. Autran, and M. Houssa, "Frequency Characterization and Modeling of Interface Traps in HfSi_xO_y/HfO₂ Gate Dielectric Stack from A Capacitance Point-of-View," *Appl. Phys. Lett.*, vol. 81, pp. 3392, 2002.

- [17] T. Kauerauf, R. Degraeve, E. Cartier, C. Soens and G. Groeseneken, "Low Weibull Slope of Breakdown Distributions in High-k Layers," *IEEE Electron Device Lett.*, vol. 23, pp. 215, 2002.
- T. Kauerauf, R. Degraeve, E. Cartier, B. Bovoreanu, P. Blomme, B. Kaczer,
 L. Pantisano, A. Kerber, and G. Groeseneken, "Towards Understanding Degradation and Breakdown on SiO₂/High-k Stack," in *IEDM Tech. Dig.*, pp. 521, 2002.
- [19] R. Degraeve, T. Kauerauf A. Kerber, E. Cartier, B. Govoreanu, Ph. Roussel,
 L. Pantisano, P. Blomme, B. Kaczer, G. Groeseneken, "Stress Polarity
 Dependence of Degradation and Breakdown of SiO₂/High-k Stacks," in *Proc. Int. Reliability Physics Symp.*, pp. 23, 2003.
- [20] A.L.P. Rotondaro, M.R. Visokay, A. Shanware, J.J. Chambers, and L. Colombo, "Carrier Mobility in MOSFETs Fabricated with Hf-Si-O-N Gate Dielectric, Poly-Silicon Gate Electrode, and Self-aligned Source and Drain," *IEEE Electron Device Lett.*, vol. 23, pp. 603, 2002.
- [21] W.J. Zhu, T.P. Ma, S. Zafar, and T. Tamagawa, "Charge Trapping in Ultrathin Hafnium Oxide," *IEEE Electron Device Lett.*, vol. 23, pp. 597, 2002.
- Y.H. Kim, K. Onishi, C.S. Kang, H-J. Cho, R. Nieh, S. Gopalan, R. Choi, J. Han, S. Krishnan, and J.C. Lee, "Area Dependence of TDDB Characteristics for HfO₂ Gate Dielectrics," *IEEE Electron Device Lett.*, vol. 23, pp. 594, 2002.

- [23] S.J. Lee, H.F. Luan, W.P. Bai, C.H. Lee, T.S. Jeon, Y. Senzaki, D. Roberts,
 D.L. Kwong, "High Quality Ultra Thin Cvd HfO₂ Gate Stack with Poly-Si
 Gate Electrode," in *IEDM Tech. Dig.*, pp. 31, 2000.
- [24] C-W. Yang, Y-K. Fang, C-H. Chen, W-D. Wang, T-Y. Lin, M-F. Wang, T-H. Hou, J-Y. Cheng, L-G. Yao, S-C. Chen, C-H. Yu, M-S. Liang, "Dramatic Reduction of Gate Leakage Current in 1.61 nm HfO₂ High-k Dielectric Poly-Silicon Gate with Al₂O₃ Capping Layer," *IEEE Electron. Lett.*, vol. 38, pp. 1223, 2002.
- [25] C.E. Weintraub, "Investigation of Charge Pumping Techniques for Advanced Gate Dielectrics," *Department of Electrical and Computer Engineering, PhD. Thesis*, North Carolina State University, 2000.
- [26] B. Ho, T. Ma, S.A. Campbell, and W.L. Gladfelter, "A 1.1 nm Oxide Equivalent Gate Insulator Formed Using TiO₂ on Nitrided Silicon," in *IEDM Tech. Dig.*, pp. 1038, 1998.
- [27] E.M. Vogel, W.L. Hill, V. Misra, P.K. McLarty, J.J. Wortman, J.R. Hauser, P. Morfouli, G. Ghibaudo, and T. Ouisse, "Mobility Behavior of n-Channel and p-Channel MOSFET's with Oxynitride Gate Dielectrics Formed by Low-Pressure Rapid Thermal Chemical Vapor Deposition," *IEEE Trans. Electron Devices*, vol. 43, pp. 753, 1996.
- [28] W.L. Hill, E.M. Vogel, V. Misra, P.K. McLarty, and J.J> Wortman, "Low-Pressure Rapid Thermal Chemical Vapor Deposition of Oxynitride Gate dielectrics for n-Channel and p-Channel MOSFET's *IEEE Trans. Electron Devices*, vol. 43, pp. 15, 1996.

- [29] V. Misra, H. Lazar, Z. Wang, and Y. Wu, "Interfacial Properties of Ultrathin Pure Silicon Nitride Formed by Remote Plasma Enhanced Chemical Vapor Deposition," *J. Vac. Sci. Technol. B*, vol. 17, pp.1836, 1999.
- [30] V. Misra, H. Lazer, M. Kulkarni, Z. Wang, G. Lucovsky, and J.R. Hauser, "Interfacial Properties of Si-Si3N4 formed by Remote Plasa Enhanced Chemical Vapor Deposition," *Mat. Res. Soc. Symp. Proc.*, vol. 567, pp. 89, 1999.
- [31] A.S. Oates, "Reliability Issues for High-k Gate Dielectrics," in *IEDM tech*. *Dig.*, pp. 38.2.1, 2003.
- [32] C.W. Yang, Y.K. Fang, C.H. Chen, S.F. Chen, C.Y. Lin, C.S. Lin, M.F. Wang, Y.M. Lin, T.H. Hou, C.H. Chen, L.G. Yao, S.C. Chen, and M.S. Liang, "Effect of Polycrystalline-Silicon Gate Types on the Opposite Flatband Voltage Shift in *n*-Type and *p*-Type Metal–Oxide–Semiconductor Field-Effect Transistors for High-*k*-HfO₂ Dielectric," *Appl. Phys. Lett.*, vol. 83, pp. 308, 2003.
- [33] J.C. Wang, S.H. Chiao, C.L. Lee, T.F. Lei, Y.M. Lin, M.F. Wang, S.C. Chen, C.H. Yu, and M.S. Liang, "A Physical Model for the Hysteresis Phenomenon of the Ultrathin ZrO₂ Film," *J. Appl. Phys.*, vol. 92, pp. 3936-3940, 2002.
- [34] R. Degraeve, G. Groeseneken, R. Bellens, M. Depas, and H.E. Maes, "A Consistent Model for the Thickness Dependence of Intrinsic Breakdown in Ultra-thin Oxides," in *IEDM Tech. Dig.*, pp. 866, 1995.
- [35] J.H. Stathis, "Percolation Models for Gate Oxide Breakdown," J. Appl. Phys., vol. 86, pp. 5757, 1999.

- [36] J.H. Stathis, "Reliability Limits for the Gate Insulator in CMOS Technology," *IBM J. RES. & DEV.*, vol. 46, pp. 265, 2002.
- [37] E. Anolick and G. Nelson, "Low Field Time Dependent Dielectric Integrity," in *Proc. Int. Reliability Physics Symp.*, vol. 17, pp. 8, 1979.
- [38] D. Crook, "Method of Determining Reliability Screens for Time Dependent Dielectric Breakdown," in *Proc. Int. Reliability Physics Symp.*, vol. 17, pp. 1, 1979.
- [39] A. Berman, "Time-Zero Dielectric Reliability Test by A Ramp Method," in Proc. Int. Reliability Physics Symp., vol. 19, pp. 204, 1981.
- [40] J.W. McPherson and H.C. Mogul, "Underlying Physics of the Thermochemical E Model in Describing Low-Field Time-Dependent Dielectric Breakdown in SiO₂ Thin Films," J. Appl. Phys. vol. 84, pp. 1513, 1998.
- [41] J.W. McPherson and R.B. Khamankar, "Molecular Model for Intrinsic Time-Dependent Dielectric Breakdown in SiO₂ Dielectrics and the Reliability Implications for Hyper Thin Gate Oxide," *Semicond. Sci. Technol.*, vol. 15, pp. 462, 2000.
- [42] J.W. McPherson and R.B. Khamandar, "Disturbed Bonding States in SiO₂ Thin-Films and Their Impact on Time-Dependent Dielectric Breakdown," in *Proc. Int. Reliability Physics Symp.*, vol. 36, pp. 47, 1998.
- [43] D.J. DiMaria, "Defect Generation Under Substrate-Hot-Electron Injection into Ultrathin Silicon Dioxide Layers," J. Appl. Phys., vol. 86, pp. 2100, 1999.

- [44] R. Degraeve, G. Groeseneken, I.D. Wolf, and H.E. Maes, "Oxide and Interface Degradation and Breakdown under Medium and High filed Injection Conditions: A Correlation Study," *Microelectron. Eng.*, vol. 28, pp. 313, 1995.
- [45] K. Umeda, T. Tmoita, and K. Taniguchi, "Silicon Dioxide Breakdown Induced by SHE (Substrate Hot Electron) Injection," *Electron. Commun. Jpn.*, pt. 2, vol. 80, pp. 11, 1997.
- [46] E.M. Vogel, J.S. Suehle, M.d. Edelstein, B. Wang, Y. chen, and J.B. Bernstein, "Reliability of Ultrathin Silicon Dioxide Under Combined Substrate Hot-Electron and Constant Voltage Tunneling Stress," IEEE Trans. Electron Devices, vol. 47, pp. 1183, 2000.
- [47] C. Chen, S. E. Holland, K. K. Young, C. Chang, and C. Hu, "Substrate hole current and oxide breakdown," *Appl. Phys. Lett.*, vol. 74, p. 669, 1986.
- [48] K.F. Schuegraf and C. Hu, "Metal-Oxide-Semiconductor Field-Effect-Transistor Substrate Current During Fowler-Nordheim Tunneling Stress and Silicon Dioxide Reliability," J. Appl. Phys., vol. 76, pp. 3695, 1994.
- [49] K.F. Schuegraf and C. Hu, "Effects of Temperature and Defects on Breakdown Lifetime of Thin SiO₂ at Very Low Voltages," in *Proc. Int. Reliability Physics Symp.*, pp. 126, 1994.
- [50] R. Degraeve, G. Groeseneken, R. Bellens, J.L. Ogier, M. Depas, P.J. Roussel, and H.E. Maes, "New Insights in The Relation Between Electron Trap Generation and the Statistical Properties of Oxide Breakdown," *IEEE Trans. Electron Devices*, vol. 45, pp. 904, 1998.

- [51] B. Weir, M. Alam, J. Bude, P. Silverman, A. Ghetti, F. Baumann, P. Diodato,
 D. Monroe, T. Sorsch, G. Timp, Y. Ma, M. Brown, A. Hamad, D. Hwang, and
 P. Mason, "Gate Oxide Reliability Projection to the Sub-2-nm Regime," *Semicond. Sci. Technol.*, vol. 15, pp. 455, 2000.
- [52] J.D. Bude, B.E. Weir, and P.J. Silverman, "Explanation of Stress-Induced Damage in Thin Oxides," in *IEDM Tech. Dig.*, pp.179, 1998.
- [53] C. Chen, S.E. Holland, K.K. Young, C. Chang, and C. Hu, "Substrate Hole Current and Oxide Breakdown," *Appl. Phys. Lett.*, vol. 74, pp. 669, 1986.
- [54] P. Samanta, and C.K. Sarkar, "Correlation Between the Gate Bias Dependence of the Probability of Anode Hole Injection and Breakdown in Thin Silicon Dioxide Films," *Appl. Phys. Lett.*, vol. 77, pp. 4350, 2000.
- [55] D.J. Dimaria, "Explanation for the Polarity Dependence of Breakdown in Ultrathin Silicon Dioxide Films," *Appl. Phys. Lett.*, vol. 68, pp. 3004, 1996.
- [56] E. Wu, and J. Suñé, "New Insights in Polarity-Deendent Oxide Breakdown for Ultrathin Gate Oxide," *IEEE Electron Device Lett.*, vol. 23, pp. 494, 2002.
- [57] M.A. Alam, J. Bude, and A. Ghetti, "Field Acceleration for Oxide Breakdown-Can An Accurate Anode Hole Injection Model Resolve the E vs. 1/E Controversy?," in *Proc. Int. Reliability Physics Symp.*, pp. 21, 2000.
- [58] H. Satake, S. Takagi, and A. Toriumi, "Evidence of Electron–Hole Cooperation in SiO₂ Dielectric Breakdown," in *Proc. Int. Reliability Physics Symp.*, pp.156, 1997.

- [59] H. Satake and A. Toriumi, "Temperature Dependent Hole Fluence to Breakdown in Thin Gate Oxides Under Fowler–Nordheim Electron Tunneling Injection," *Appl. Phys. Lett.*, vol. 66, pp. 3516, 1995.
- [60] E.M. Vogel, M.D. Edelstein, and J.S. Suehle, "Defect Generation and Breakdown of Ultrathin Silicon Dioxide Induced by Substrate Hot-Hole Injection," J. Appl. Phys. vol. 90, pp.2338, 2001.
- [61] D.J. DiMaria, "Anode Hole Injection and Trapping in Silicon Dioxide," J. Appl. Phys., vol. 80, pp. 304, 1996.
- [62] M. Rasras, I.D. Wolf, G. Groeseneken, B. Kaczer, R. Degraeve, and H.E. Maes, "Photo-Carrier Generation As the Origin of Fowler–Nordheim Induced Substrate Hole Current in Thin Oxides," in *IEDM Tech. Dig.*, pp. 465, 1999.
- [63] D.J. DiMaria and J.W. Stasiak, "Trap Creation in Silicon Dioxide Produced by Hot Electrons," J. Appl. Phys., vol. 65, pp. 2342, 1989.
- [64] A. Yokozawa, and Y. Miyamoto, "Hydrogen Dynamics in SiO₂ Triggered by Electronic Excitations," J. Appl. Phys., vol. 88, pp. 4542, 2000.
- [65] T-C. Shen, C. Wang, G.C. Abeln, J.R. Tucker, J.W. Lyding, P. Avouris, and R.E. Walkup, "Atomic Scale Desorption Through Electronic and Vibrational Excitation Mechanisms Science," *Science*, vol. 268, pp. 1590, 1995.
- [66] E.Y. Wu, A. Vayshenker, E. Nowak, J. Suñé, R-P. Vollertsen, W. Lai, and D. Harmon, " Experimental Evidence of T_{BD} Power-Law for Voltage Dependence of Oxide Breakdown in Ultrathin Gate Oxides," *IEEE Trans. Electron Devices*, vol. 49, pp. 2244, 2002.

- [67] W. McMahon, A. Haggag, and K. Hess, "Reliability Scaling Issues for Nanoscale Devices," *IEEE Trans. Nanotechnol.*, vol. 2, pp. 33, 2003.
- [68] J.Wu, E. Rosenbaum, B. MacDonald, J.T.E. Li, B. Tracy, and P. Fang,
 "Anode Hole Injection Versus Hydrogen Release: The Mechanism for Gate
 Oxide Breakdown," in *Proc. Int. Reliability Physics Symp.*, pp. 27, 2000.
- [69] K. Hess, I.C. Kizilyalli, and J.W. Lyding, "Giant Isotope Effect in Hot Electron Stress of Metal Oxide Silicon Devices," *IEEE Trans. Electron Devices*, vol. 45, pp. 406, 1998.
- [70] I. C. Kizilyalli et al., "Improvement of Hot Carrier Reliability with Deuterium Anneals for Manufacturing Multilevel Metal/Dielectric MOS Systems," *IEEE Electron. Dev. Lett.*, vol. 19, pp. 444, 1998.
- [71] W. F. Clark et al., "Process Stability of Deuterium-Annealed MOSFET's," *IEEE Electron. Dev. Lett.*, vol. 20, pp. 48-50, 1999.
- [72] C.E. Ebeling, An Introduction to Reliability and Maintainability Engineering.McGraw-Hill, 1997.
- [73] R. Degraeve, J.L. Ogier, R. Bellens, P. Roussel, G. Groeseneken, and H.E.
 Maes, "On the Field Dependence of Intrinsic and Extrinsic Time-dependent
 Dielectric Breakdown," in *Proc. Int. Reliability Physics Symp.*, pp. 44, 1996.
- [74] E. Wu, J.H. Stathis, and L-K. Han, "Ultra-thin Oxide Reliability for ULSI applications," *Semicond. Sci. Technol.*, vol. 15, pp. 425, 2000.
- [75] R. Degraeve, B. Kaczer, and G. Groeseneken, "Reliability: A Possible Showstopper for Oxide Thickness Scaling?" *Semicond. Sci. Technol.*, vol. 15, pp. 436, 2000.

- [76] E. Wu and R-P. Vollertsen, "On the Weibull Shape Factor of Intrinsic Breakdown of Dielectric Films and Its Accurate Experimental Determination -- Part I: Theory, Methodology, Experimental Techniques," *IEEE Trans. Electron Devices*, vol. 49, pp. 2131, 2002.
- [77] E. Wu and R-P. Vollertsen, "On the Weibull Shape Factor of Intrinsic Breakdown of Dielectric Films and Its Accurate Experimental Determination -- Part II: Experimental Results and the Effects of Stress Conditions," *IEEE Trans. Electron Devices*, vol. 49, pp. 2141, 2002.
- [78] J.W. McPherson, V. Reddy, K. Banerjee, and H. Le, "Comparison of E and 1/E TDDB Modes for SiO2 under Long-term/Low-field Test Conditions," in *IEDM Tech. Dig.*, pp. 171, 1998.
- [79] J.S. Suehle, "Ultrathin Gate Oxide Reliability: Physical Models, Statistics, and Characterization," *IEEE Trans. Electron Devices*, vol. 49, pp. 958, 2002.
- [80] E. Wu, E.J. Nowak, R–P. Vollertsen, and L–K. Han, "Weibull Breakdown Characteristics and Oxide Thickness Uniformity," *IEEE Trans. Electron Devices*, vol. 47, pp. 2301, 2000.
- [81] E.Y. Wu, W.W. Abadeer, L-K. Han, S.H. Lo, and G. Hueckel, "Challenges for Accurate Reliability Projections in the Ultra-thin Oxide Regime," in *Proc. Int. Reliability Physics Symp.*, vol. 37, pp. 57, 1999.
- [82] B.P. Linder, J.H. Stathis, and D.J. Frank, "Calculating the Error in Long Term Oxide Reliability Estimates," in *Proc. Int. Reliability Physics Symp.*, pp. 168, 2001.

- [83] J.G. Simmons and L.S. Wei, "Theory of Dynamic Charge Current and Capacitance Characteristics in Mis Systems Containing Distributed Surface Traps," *Solid-State Electronics*, vol. 16, pp. 53, 1973.
- [84] G. Groeseneken, H.E. Maes, N. Beltran, and R.F. Dekeersmaecker, "A Reliable Approach to Charge-pumping Measurements in Mos Transistors," *IEEE Trans. Electron Devices*, vol. 31, pp. 42, 1984.
- [85] P. Heremans, J. Witters, G. Groeseneken, and H.E. Maes, "Analysis of the Charge Pumping Technique and Its Application for the Evaluation of MOSFET Degradation," *IEEE Trans. Electron Devices*, vol. 36, pp. 1318, 1989.
- [86] Y. Maneglia and D. Bauza, "Extraction of Slow Oxide Trap Concentration Profiles in Metal-oxide-semiconductor Transistors Using the Charge Pumping Method," J. Appl. Phys., vol. 79, pp. 4187, 1996.
- [87] J.L. Autran, and C. Chabrerie, "Use of the Charge Pumping Technique with A Sinusoidal Gate Waveform," *Solid-State Electronics*, vol. 39, pp. 1394, 1996.
- [88] G. Ghibaudo and N.S. Saks, "A Time Domain Analysis of the Charge Pumping Current," J. Appl. Phys., vol. 64, pp. 4751, 1988.
- [89] G. Ghibaudo and N.S. Saks, "Investigation of the Charge Pumping Current in Metal-oxide-semiconductor Structures," J. Appl. Phys., vol. 65, pp. 4311, 1989.
- [90] N.S. Saks, "Measurement of Single Interface Trap Capture Cross Sections with Charge Pumping," *Appl. Phys. Lett.*, vol. 70, pp. 3380, 1997.

- [91] D.J. DiMaria and E. Cartier, "Mechanism for Stress-induced Leakage Currents in Thin Silicon Dioxide Films," J. Appl. Phys., vol. 78, pp. 3883, 1995.
- [92] A. Yokozawa, A. Oshiyama, Y. Miyamoto, and S. Kumashiro, "Oxygen Vacancy with Large Lattice Distortion as An Origin of Leakage Currents in SiO₂," in *IEDM Tech. Dig.*, pp. 703, 1997.
- [93] D.A. Buchanan, D.J. DiMaria, C-A. Chang, and Y. Taur, "Defect Generation in 3.5 nm Silicon Dioxide Films," *Appl. Phys. Lett.*, vol. 65, pp. 1820, 1994.
- [94] M.A. Alam, "SILC as A Measurement of Trap Generation and Predictor of T_{BD} in Ultrathin Oxides," *IEEE Trans. Electron Devices*, vol. 49, pp. 226, 2002.
- [95] D. Ielmini, A.S. Spinelli, A.L. Lacaita, and G. Ghidini, "Evidence for Recombination at Oxide Defects and New SILC Model," in *Proc. Int. Reliability Physics Symp.*, pp. 55, 2000.
- [96] C. Chen, S.E. Holland, K.K. Young, C. Chang, and C. Hu, "Substrate Hole Current and Oxide Breakdown," *Appl. Phys. Lett.*, vol. 74, pp. 669, 1986.
- [97] J.H. Stathis and D.J. DiMaria, "Reliability Projection for Ultra-Thin Oxides at Low Voltage," in *IEDM Tech. Dig.*, pp. 167, 1998.
- [98] M.A. Alam, B. Weir, J. Bude, P. Silverman, D. Monroe, "Explanation of Soft and Hard Breakdown and Its Consequences for Area Scaling," in *IEDM Tech. Dig.*, pp. 449, 1999.
- [99] D.J. DiMaria and J.W. Stasiak, "Trap Creation in Silicon Dioxide Produced by Hot Electrons," J. Appl. Phys., vol. 65, pp. 2342, 1989.

- [100] D.J. DiMaria, D. Arnold, and E. Cartier, "Impact Ionization and Positive Charge Formation in Silicon Dioxide Films on Silicon," *Appl. Phys. Lett.*, vol. 60, pp. 2118, 1992.
- [101] G. Groeseneken, R. Degraeve, T. Nigam, G.V. D. Bosch, and H.E. Maes, "Hot Carrier Degradation and Time-Dependent Dielectric Breakdown in Oxides," *Microelectron. Eng.*, vol. 49, pp. 27, 1999.
- [102] R. Degraeve, G. Groeseneken, R. Bellens, M.Depas, and H. E. Maes, "A Consistent Model for the Thickness Dependence of Intrinsic Breakdown in Ultra-Thin Oxides," in *IEDM Tech. Dig.*, pp. 863, 1995.
- [103] J.H. Stathis, "Percolation Models for Gate Oxide Breakdown," J. Appl. Phys., vol. 86, pp. 5757, 1999.
- [104] E.M. Vogel, M.D. Edelstein, and J.S. Suehle, "Defect Generation and Breakdown of Ultrathin Silicon Dioxide Induced by Substrate Hot-Hole Injection," J. Appl. Phys. vol. 90, pp.2338, 2001.
- [105] D.J. DiMaria, "Electron Energy Dependence of Metal-Oxide-Semiconductor Degradation," *Appl. Phys. Lett.* vol. 75, pp. 2427, 1999.
- [106] D. Esseni, J. D. Bude and L. Selmi, "On Interface and Oxide Degradation in VLSI MOSFETs. I. Deuterium Effect in CHE Stress Regime," *IEEE Trans. Electron Devices*, vol. 49, pp. 254, 2002.
- [107] M. A. Alam, "SILC as A Measure of Trap Generation and Predictor of T_{BD} in Ultrathin Oxides," *IEEE Trans. Electron Devices*, vol. 49, pp. 226, 2002.
- [108] P. Heremans, J. Witters, G. Groeseneken, and H.E. Maes, "Analysis of the Charge Pumping Technique and Its Application for the Evaluation of

MOSFET Degradation," *IEEE Trans. Electron Devices*, vol. 36, pp. 1318, 1989.

- [109] G. Groeseneken, H.E. Maes, N. Beltran, and R.F. Dekeersmaecker, "A Reliable Approach to Charge-pumping Measurements in Mos Transistors," *IEEE Trans. Electron Devices*, vol. 31, pp. 42, 1984.
- [110] S.K. Lai, "Interface Trap Generation in Silicon dioxide When Electrons Are Captured by Trapped Holes," J. Appl. Phys. vol. 54, pp. 2540, 1983.
- [111] B.P. Linder, J.H. Stathis, and D.J. Frank, "Calculating the Error in Long Term Oxide Reliability Estimates," in *Proc. Int. Reliability Physics Symp.*, pp. 168, 2001.
- [112] J.H. Stathis *et al.*, "Breakdown Measurements of Ultra-Thin SiO₂ at Low Voltage," 2000 VLSI Tech. Digests, pp. 94, 2000.
- [113] J.H. Stathis and D.J. DiMaria, "Reliability Projection of Ultrathin Oxides at Low Voltages," 1998 IEDM Tech. Digests, pp. 167, 1998.
- [114] M.A. Alam, "SILC as A Measure of Trap Generation and Predictor of T_{BD} in Ultrathin Oxides," *IEEE Trans. Electron. Devices*, vol. 49, pp. 226, 2002.
- [115] D.J. DiMaria, "Defect Generation Under Substrate-Hot-Electron Injection into Ultrathin Silicon Dioxide Layers," J. Appl. Phys. vol. 86, pp. 2100, 1999.
- [116] P. Heremans, J. Witters, G. Groeseneken, and H.E. Maes, "Analysis of the Charge Pumping Technique and Its Application for the Evaluation of MOSFET Degradation," *IEEE Trans. Electron Devices*, vol. 36, pp. 1318, 1989.

- [117] G. Groeseneken, H.E. Maes, N. Beltran, and R.F. De Keersmaecker, "A Reliable Approach to Charge-Pumping Measurements in MOS Transistors," *IEEE Trans. Electron Devices*, vol. 4, pp. 42, 1984.
- [118] D. Esseni, J.D. Bude, and L. Selmi, "On Interface and Oxide Degradation in VLSI MOSFETs. II. Fowler-Nordheim Stress Regime," *IEEE Trans. Electron Devices*, vol. 49, pp. 254, 2002.
- [119] R.N. HALL, "Electron-Hole Recombination in Germanium," *Phys. Rev.*, vol. 87, pp. 387, 1952.
- [120] W. Schockley and W.T. Read, "Statistics of the Recombination of Hole and Electrons," *Phys. Rev.*, vol. 87, pp. 835, 1952.
- [121] B. Wang, J.S. Suehle, E.M. Vogel and J.B. Bernstein, "Time-Dependent Breakdown of Ultra-Thin SiO₂ Gate Dielectrics Under Pulsed Biased Stress," *IEEE Electron Device Lett.*, vol. 22, pp. 224, 2001.
- [122] R.M. PATRIKAR, R. LAL, and J. VASI, "Degradation of Oxides in Metal-Oxide-Semiconductor Capacitors Under High-Field Stress," J. Appl. Phys., vol. 74, pp. 4598, 1993.
- [123] C. Hu, S.C. Tam, F. Hsu, P. Ko, T. Chan, and K.W. Terrill, "Hot-Electron-Induced MOSFET Degradation-Model, Monitor, and Improvement," *IEEE Trans. Electron. Devices*, vol. 32, pp. 375, 1985.
- [124] C.T. Sune, A. Reisman, and C.K. Williams, "A New Electron-Trapping Model for the Gate Insulator of Insulated Gate Field-Effect Transistors," J. *Electron. Mat.*, vol. 19, pp. 651, 1990.

- [125] J.L. Autran, and C. Chabrerie, "Use of the Charge Pumping Technique with A Sinusoidal Gate Waveform," *Solid-St. Electron.*, vol. 39, pp. 1394, 1996.
- [126] J-S. Lee, J.W. Lyding, and K. Hess, "Hydrogen-Related Extrinsic Oxide Trap Generation in Thin Gate Oxide Film During Negative-Bias Temperature Instability Stress," in *Proc. Int. Reliability Physics Symp.*, pp. 685, 2004.
- [127] D. Heh, E.M. Vogel, and J. B. Bernstein, "Impact of Substrate Hot Hole Injection on Ultrathin Silicon Dioxide Breakdown," *Appl. Phys. Lett.*, vol. 82, pp. 3242, 2003.
- [128] E.Y. Wu, A. Vayshenker, E. Nowak, J. Suñé, R-P. Vollertsen, W. Lai, and D. Harmon, "Experimental Evidence of T_{BD} Power-Law for Voltage Dependence of Oxide Breakdown in Ultrathin Gate Oxides," *IEEE Trans. Electron Devices*, vol. 49, pp. 2244, 2002.
- [129] P. Moens, G. Van den bosch, and G. Groeseneken, "Hot-Carrier Degradation Phenomena in Lateral and Vertical DMOS Transistors," *IEEE Trans. Electron Devices*, vol. 51, pp. 623, 2004.
- [130] D.S. Ang and C.H. Ling, "On the Dominant Interface Trap Generation Process During Hot-Carrier Stressing [MOSFETs]," in *Proc. Int. Reliability Physics Symp.*, pp. 412, 2001.
- [131] F. Irrera and B. Ricco, "SILC Dynamics in MOS Structures Subject to Periodic Stress," *IEEE Trans. Electron Devices*, vol. 49, pp. 1729, 2002.
- [132] S. Mahapatra, K.P.B. Kumar, and M.A. Alam, "Investigation and Modeling of Interface and Bulk Trap Generation During Negative Bias Temperature

Instability of p-MOSFETs," *IEEE Trans. Electron Devices*, vol. 51, pp. 1371, 2004.

- [133] G. Chen, M.F. Li, and Y. Jin, "Interaction of Interface-Traps Located at Various Sites in MOSFETs Under Stress," *IEEE Trans. Reliability*, vol. 51, pp. 387, 2002.
- [134] Y. Tang, Y. Hao, J. Zhu, and J. Zhang, "Degradation Model of the Electron Gate Current in PMOSFET," in *Proc. Int. Solid-State and Integrated-Circuit Technology*, vol. 2, pp. 1014, 2001.
- [135] M. Xu, C. Tan, H. Chen, and X. Duan, "Current Induced Subthreshold Trap Generation, Degradation, and Breakdown in the Thin Oxide," in *Proc. Int. Solid-State and Integrated-Circuit Technology*, vol. 2, pp. 932, 2001.
- [136] O. Penzin, A. Haggag, W. McMahon, E. Lyumkis, and K. Hess, "MOSFET Degradation Kinetics and Its Simulation," *IEEE Trans. Electron Devices*, vol. 51, pp. 1445, 2003.
- [137] M.A. Alam, "A Critical Examination of the Mechanics of Dynamic NBTI for PMOSFETs," in *IEDM Tech. Dig.*, pp. 14.4.1, 2003.
- [138] J.F. Zhang, C.Z. Zhao, A.H. Chen, G. Groeseneken, and R. Degraeve, "Hole Traps in Silicon Dioxides. Part I. Properties," *IEEE Trans. Electron Devices*, vol. 51, pp. 1267, 2004.
- [139] J.F. Zhang, C.Z. Zhao, H.K. Sii, G. Groeseneken, R. Degraeve, J.N. Ellis, and C.D. Beech, "Relation Between Hole Traps and Hydrogenous Species in Silicon Dioxides," *Solid-State Electron.*, vol. 46, pp. 1839, 2002.

- [140] Y. Toyoshima, H. Iwai, F. Matsuoka, H. Hayashida, K. Maeguchi, and K. Kanzaki, "Analysis on Gate-Oxide Thickness Dependence of Hot-Carrier-Induced Degradation in Thin-Gate Oxide nMOSFET's," *IEEE Trans. Electron Devices*, vol. 37, pp. 1496, 1990.
- [141] M.S. Krishnan, Y.C. Yeo, Q. Lu, T-J. King, J. Bokor, and C. Hu, "Remote Charge Scattering in MOSFETs with Ultra-Thin Gate Dielectrics," in *IEDM Tech. Dig.*, pp. 571, 1998.
- [142] J.S. Suehle, "Ultrathin Gate Oxide Reliability: Physical Models, Statistics, and Characterization," *IEEE Trans. Electron Devices*, vol. 49, pp. 958-971, 2002.
- [143] E. Wu, S.H. Lo, W. Abadeer, A. Acovic, D. Buchanan, T. Furukawa, D. Brochu, and R. Dufresne, "Determination of Ultra-Thin Oxide Voltages and Thickness and the Impact on Reliability Projection," in *Proc. Int. Reliability Physics Symp.*, pp. 184-189, 1997.
- [144] G. Timp et al., "Low Leakage, Ultra-Thin Gate Oxides for Extremely High Performance Sub-100 nm nMOSFETs," in *IEDM Tech. Dig.*, pp. 930-932, 1998.
- [145] J.H. Stathis and D.J. DiMaria, "Reliability Projection for Ultra-Thin Oxides at Low Voltage," in *IEDM Tech. Dig*, pp. 167-170, 1998.
- [146] Y. Taur and E.J. Nowak, "CMOS Devices below 0.1 um: How High Will Performance Go," in *IEDM Tech. Dig.*, pp. 215-218, 1997.
- [147] S.H. Lo, D.A. Buchanan, Y. Taur, and W. Wang, "Quantum-Mechanical Modeling of Electron Tunneling Current from the Inversion Layer of

Ultrathin-Oxide nMOSFETs," *IEEE Electron Device Lett.*, vol. 18, pp. 209-211, 1997.

- [148] E.M. Vogel, D. Heh, and J.B. Bernstein, "Interaction between Low-Energy Electrons and Defects Created by Hot Holes in Ultrathin Silicon Dioxide," *Appl. Phys. Lett.*, vol. 80, pp. 3343-3345, 2002.
- [149] E.M. Vogel, D. Heh, J.B. Bernstein, and J.S. Suehle, "Impact of the Trapping of Anode Hot Holes on Silicon Dioxide Breakdown," *IEEE Electron Device Letters*, vol. 23, pp. 667-669, 2002.
- [150] D. Heh, E.M. Vogel, and J.B. Bernstein, "Impact of Substrate Hot Hole Injection on Ultrathin Silicon Dioxide Breakdown," *Appl. Phys. Lett.*, vol. 82, pp. 3242-3244, 2003.
- [151] G-W. Lee, J-H. Lee, H-W. Lee, M-K Park, D-G. Kang, and H-K Youn, "Trap Evaluations of Metal/Oxide/Silicon Field-effect Transistors with High-k Gate Dielectric Using Charge Pumping Method," *Appl. Phys. Lett.*, vol. 81, pp. 2050-2052, 2002.
- [152] S. Zafar, A. Callegari, E. Gusev, M.V. Fischetti, "Charge Trapping in High-k Gate Dielectric Stacks," in *IEDM Tech. Dig.*, pp. 517-520, 2002.
- [153] P. Masson, J.-L. Autran, M. Houssa, X. Garros, and C. Leroux, "Frequency Characterization and Modeling of Interface Traps in HfSi_xO_y/HfO₂ Gate Dielectric Stack from A Capacitance Point-of-View," *Appl. Phys. Lett.*, vol. 81, pp. 3392-3394, 2002.

- [154] T. Kauerauf, R. Degraeve, E. Cartier, C. Soens and G. Groeseneken, "Low Weibull Slope of Breakdown Distributions in High-k Layers," *IEEE Electron Device Lett.*, vol. 23, pp. 215-217, 2002.
- T. Kauerauf, R. Degraeve, E. Cartier, B. Bovoreanu, P. Blomme, B. Kaczer,
 L. Pantisano, A. Kerber, and G. Groeseneken, "Towards Understanding Degradation and Breakdown on SiO₂/High-k Stack," in *IEDM Tech. Dig.*, pp. 521-524, 2002.
- [156] R. Degraeve, T. Kauerauf A. Kerber, E. Cartier, B. Govoreanu, Ph. Roussel,
 L. Pantisano, P. Blomme, B. Kaczer, G. Groeseneken, "Stress Polarity
 Dependence of Degradation and Breakdown of SiO₂/High-k Stacks," in *Proc. Int. Reliability Physics Symp.*, pp. 23-28, 2003.
- [157] A.L.P. Rotondaro, M.R. Visokay, A. Shanware, J.J. Chambers, and L. Colombo, "Carrier Mobility in MOSFETs Fabricated with Hf-Si-O-N Gate Dielectric, Poly-Silicon Gate Electrode, and Self-aligned Source and Drain," *IEEE Electron Device Lett.*, vol. 23, pp. 603-605, 2002.
- [158] W.J. Zhu, T.P. Ma, S. Zafar, and T. Tamagawa, "Charge Trapping in Ultrathin Hafnium Oxide," *IEEE Electron Device Lett.*, vol. 23, pp. 597-599, 2002.
- [159] Y.H. Kim, K. Onishi, C.S. Kang, H-J. Cho, R. Nieh, S. Gopalan, R. Choi, J. Han, S. Krishnan, and J.C. Lee, "Area Dependence of TDDB Characteristics for HfO₂ Gate Dielectrics," *IEEE Electron Device Lett.*, vol. 23, pp. 594-596, 2002.
- [160] J-P. Han, E.M. Vogel, E.P. Gusev, C. D'Emic, C.A. Richter, D. Heh, and J.S. Suehle, "Asymmetric Energy Distribution of Interface Traps in n- and p-

MOSFETs with HfO₂ Gate Dielectric on Ultrathin SiON buffer layer," *IEEE Electron Device Letters*, vol. 25, pp. 126-128, 2004.

- [161] J. F. Jr. Conley, Y. Ono, W. Zhuang, L. Stecker, and G. Stecker, "Electrical Properties and Reliability of HfO₂ Deposited via ALD Using Hf(NO₃)₄
 Precursor," in *Integrated Reliability Workshop final report*, pp. 108-112, 2002.
- [162] C.W. Yang, Y.K. Fang, C.H. Chen, S.F. Chen, C.Y. Lin, C.S. Lin, M.F. Wang, Y.M. Lin, T.H. Hou, C.H. Chen, L.G. Yao, S.C. Chen, and M.S. Liang, "Effect of Polycrystalline-Silicon Gate Types on the Opposite Flatband Voltage Shift in *n*-Type and *p*-Type Metal–Oxide–Semiconductor Field-Effect Transistors for High-*k*-HfO₂ Dielectric," *Appl. Phys. Lett.*, vol. 83, pp. 308-310, 2003.
- [163] J.C. Wang, S.H. Chiao, C.L. Lee, T.F. Lei, Y.M. Lin, M.F. Wang, S.C. Chen, C.H. Yu, and M.S. Liang, "A Physical Model for the Hysteresis Phenomenon of the Ultrathin ZrO₂ Film," *J. Appl. Phys.*, vol. 92, pp. 3936-3940, 2002.
- [164] Y.-L. Chu, D.-W. Lin, and C.-Y. Wu, "A New Charge-pumping Technique for Profiling the Interface-states and Oxide-trapped Charges in MOSFETs," *IEEE Trans. Electron Devices*, vol. 47, pp. 348-353, 2000.
- S. Mahapatra, C.D. Parikh, V. Ramgopal Rao, C.R. Viswanathan, and J. Vasi,
 "A Comprehensive Study of Hot-carrier Induced Interface and Oxide Trap Distributions in MOSFETs Using A Novel Charge Pumping Technique," *IEEE Trans. Electron Devices*, vol. 47, pp. 171-177, 2000.

- [166] S. Mahapatra, C.D. Parikh, and J. Vasi, "A New "Multifrequency" Charge Pumping Technique to Profile Hot-Carrier-Induced Interface-State Density in nMOSFET's," *IEEE Trans. Electron Devices*, vol. 46, pp. 960-967, 1999.
- [167] P. Heremans, J. Witters, G. Groeseneken, and H.E. Maes, "Analysis of the Charge Pumping technique and Its Application for the Evaluation of MOSFET Degradation," *IEEE Trans. Electron Devices*, vol. 36, pp. 1318-1335, 1989.
- [168] G. Groeseneken, H.E. Maes, N. Beltran, and R. F. De keersmaecker, "A Reliability Approach to Charge-Pumping Measurements in MOS Transistors," *IEEE Trans. Electron Devices*, vol. 31, pp. 42-53, 1984.
- [169] Y. Maneglia and D. Bauza, "Extraction of Slow Oxide Trap Concentration Profiles in Metal-Oxide-Semiconductor Transistors Using the Charge Pumping Method," J. Appl. Phys., vol. 79, pp.4187-4192, 1996.
- [170] D. Bauza and Y. Maneglia, "In-Depth Exploration of Si-SiO₂ Interface Traps in MOS Transistors Using the Charge Pumping Technique," *IEEE Trans. Electron Devices*, vol. 44, pp. 2262-2266, 1997.
- [171] R.N. Hall, "Electron-Hole Recombination in Germanium," *Phys. Rev.*, vol. 87, pp. 387-387, 1952.
- [172] W. Shockley and W.T. Read, "Statistics of the Recombinations of Holes and Electrons," *Phys. Rev.*, vol. 87, pp. 835-842, 1952.
- [173] R.S. Muller and T. I. Kamins, Device Electronics for Integrated Circuits, Wiley, 1986.

- [174] F.P. Heiman, G. Warfield, "The Effects of Oxide Traps on the MOS Capacitance," *IEEE Trans. Electron Devices*, vol. 12, pp. 167-178, 1965.
- [175] J. Barnett, N. Moumen, J. Gutt, M. Gardner, C. Huffman, P. Majhi, J. J. Peterson, S. Gopalan, B. Foran, H.-J. Li, B. H. Lee, G. Bersuker, P. Zeitzoff, G. A. Brown, P. Lysaght, C. D. Young, R. W. Murto, and H. R. Huff, "Experimental Study of Etched Back Thermal Oxide for Optimization of the Si/High-k Interface," presented at 2004 Spring Meeting of the Material Research Society, p. E1.4.1, 2004.
- [176] Nelson S. Saks and Mario G. Ancona, "Determination of Interface Trap Capture Cross Sections Using Three-Level Charge Pumping," *IEEE Electron Device Letters*, vol. 11, pp. 339-341, 1990.
- [177] M. G. Ancona and N. S. Saks, "Numerical Simulation of 3-Level Charge Pumping," J. Appl. Phys., vol. 71, pp. 4415-4421, 1992.